博士論文

Entropy of the Janus interface in superconformal field theories

(超共形場理論におけるヤヌス・インターフェースのエントロピー)

永野廉人

Abstract

In this thesis we study the entropy of the Janus interface in a $4d \ \mathcal{N} = 2$ superconformal field theory (SCFT). The Janus interface is a co-dimension one defect across which a coupling constant changes its value. We show that the entropy of the Janus interface in a $4d \ \mathcal{N} = 2$ SCFT can be written by a specific linear combination of analytically continued Kähler potentials on moduli space called Calabi's diastasis.

First we give a definition of an interface entropy as a contribution from an interface to an entanglement entropy across a spherical entangling surface. Then we derive the relation between the interface entropy and the sphere partition function via the conformal map introduced by Casini, Huerta, Meyers. Finally we evaluate the entropy of the Janus interface by using this relation and SUSY localization to show that it can be written by Calabi's diastasis.

Contents

1	Intr	oductio	on	7			
	1.1	Background and motivation					
		1.1.1	Defects	7			
		1.1.2	Interface entropy	8			
		1.1.3	Janus interface and its entropy	9			
	1.2	Summa	ary of this thesis	10			
2	Con	formal	and superconformal field theory	11			
	2.1	Confor	mal Field Theory	11			
		2.1.1	Conformal symmetry and conformal algebra	11			
		2.1.2	Primary and descendant operator	13			
		2.1.3	Correlation function in conformal field theory	13			
		2.1.4	Trace anomaly	14			
	2.2	Supers	ymmetry and superconformal field theory	14			
		2.2.1	Supersymmetry	14			
		2.2.2	Superconformal symmetry	15			
	2.3	Sphere	partition function	16			
		2.3.1	$Conformal\ manifold\ .\ .\ .\ .\ .\ .\ .\ .\ .\ .\ .$	16			
		2.3.2	Divergent structure of sphere partition function	17			
		2.3.3	Supsersymmetric partition function	18			
3	Interface entropy 2						
	3.1	Entang	element entropy and sphere partition function in CFT .	22			
		3.1.1	СНМ тар	22			
		3.1.2	Original derivation of the relation	23			
		3.1.3	UV divergence	25			
		3.1.4	Another derivation	25			
	3.2	Entrop	y of a conformal interface	27			

CONTENTS

		3.2.1	Conformal interface	27
		3.2.2	Entropy of conformal interface	28
		3.2.3	UV divergence	29
	3.3	Entrop	by of a half-BPS superconformal interface	30
		3.3.1	Half-BPS superconformal interface	30
		3.3.2	Entropy of a supefconformal interface	31
		3.3.3	Supersymmetric Rényi entropy	32
4	Con	struct	ion of the Janus interface	35
	4.1	Off-she	ell Construction of Janus interface in flat space	35
		4.1.1	Step function profile	37
	4.2	Off-she	ell construction of Janus interface in \mathbb{S}^4	38
		4.2.1	Massive subalgebra on \mathbb{S}^4	38
		4.2.2	Construction	40
	4.3	Janus	interface in gauge theory	42
5	SUS	SY loca	alization and Janus interface entropy	45
	5.1	SUSY	Localization	45
	5.2	Janus	partition function	47
		5.2.1	Classical action	47
		5.2.2	Instanton and anti-instanton parition functions	49
		5.2.3	Partition function with the Janus interface	49
		5.2.4	Supergravity conterterm and Kähler ambiguity	50
	5.3	Entrop	by of Janus interface	52
	5.4	Hologr	aphic example	52
		5.4.1	The Janus solution in supergravity	52
		5.4.2	Sphere free energy	53
		5.4.3	Entanglement entropy	56
6	Con	clusio	n and outlook	59
A	Sup	ersym	metry and supergravity	61
	A.1	Notati	ons and conventions	61
		A.1.1	Signature and coordinate index	61
		A.1.2	Gamma matrix	61
		A.1.3	Charge and Weyl conjugation	62
		A.1.4	$SU(2)_R$ multiplets \ldots \ldots \ldots \ldots	63
	A.2	Supers	symmetry parameters	63
		÷		

4

	A.3	64	
		A.3.1 Vector multiplet	64
		A.3.2 Chiral multiplet	65
	A.4	Tensor calculus for chiral multiplets	65
	A.5	Definition of $\mathbb{T}(\log \overline{\Phi})$	66
	A.6	Supercurrent multiplet	67
в	Wey	vl transformation between \mathbb{S}^4 and flat space	69
С	Bou	indary super-Weyl anomaly in $2d \mathcal{N} = (2, 2)$ SCFT	71
	C.1	Super-Weyl anomaly on closed manifold	71
	C.2	Super-Weyl anomaly on open manifold and hemisphere partition	n function 72

Chapter 1

Introduction

Quantum field theories (QFTs) are very fundamental and important subjects in theoretical physics because they describe wide phenomena; particle physics, condensed matter physics, cosmology and so on. Morever QFTs themselves exhibit rich and fascinating structure. The main purpose of this thesis is to study a QFT non-perturbatively by using non-local objects called defects. In this chapter we briefly explain basic backgrounds to give motivations to the subjects of this thesis. Then we summarize main results and explain the organization of this thesis.

1.1 Background and motivation

1.1.1 Defects

What we mainly focus on in this paper are non-local objects in QFTs which are so called defects. Defects have many applications and thus are very important. We list some of them.

- The expectation value of a defect can be used as an order parameter of a phase transition. For example, a dimension one defect in quantum chromodynamics (QCD) called the Wilson loop is used to diagnose confinement.
- By inserting a defect into a QFT, we can extract data in the ambient QFT (bulk) via relations between data in the bulk and defect, *e.g.* [1].

- In a realistic setting in a laboratory, materials have boundaries and/or interfaces. This can be understood theoretically as co-dimension one defects in a QFT.
- Open string theory is formulated by a conformal field theories (CFT) on a world-sheet with a boundary.

In this paper we particularly focus on a co-dimension one defect, an interface.

1.1.2 Interface entropy

We particularly focus on an entropy associated to an interface which we call an interface entropy. Here we explain the motivation to study this quantity on a general background. A motivation to study the entropy for a specific interface (the Janus interface) will be explained in the next subsection. One of motivations to study an interface entropy is that it is conjectured to be a C-function in a CFT with an interface which we call an interface CFT or an ICFT [2]. Let me explain this point.

What is a C-function? Suppose that we have a conformal field theory (CFT) at a ultraviolet (UV) fixed point, then flow to an infrared (IR) fixed point along with a renormalization group (RG) flow. A function which decreases monotonically along with a RG flow is called a C-function. If we can find a C-function, we can get nontrivial constraints on IR quantities from UV quantities. Furthermore this function characterizes the number of degrees of freedom since the value of this function at the UV theory where many modes fluctuate is bigger than the value at the IR theory where only few modes fluctuate.

We have some examples of such functions in CFTs without an interface. In two dimensions such a function was first constructed by [3] from a particular linear combination of two point functions of the stress-energy tensor. It reduces to the coefficient of the Weyl anomaly or the *c*-central charge at fixed points. In four dimensions the *a*-coefficient of the Weyl anomaly is shown to be a weak version ¹ of *C*-function by [4]. In three dimensions it was proposed that the free energy which is the log of a partition function on a sphere is a *C*-function [5, 6]. This statement was later proved by [7].

 $^{^1{\}rm This}$ means that the value of the function at an UV fixed point is bigger than that at an IR fixed point, not necessarily decreases along RG flows.

1.1. BACKGROUND AND MOTIVATION

Then a natural question is whether we can find a C-function in ICFT. One of candidates is an entropy of an interface which we call an interface entropy [2]. So if this statement is proved we can use an interface entropy to classify/constraint theories with an interface. This motivates us to study an interface entropy.

1.1.3 Janus interface and its entropy

In the latter half of this thesis we focus on a specific interface, called the Janus interface [8, 9]. The direct motivation of our research is the fascinating relation [10] between the entropy of the Janus interface in a $2d \mathcal{N} = (2, 2)$ superconformal field theory (SCFT) and the geometric quantity on the space of CFTs called Calabi's diastasis [11].

Let us consider the space of CFTs S called the conformal manifold or the moduli space which is parametrized by moduli parameters τ^{I_2} . The Janus interface is an interface across which moduli parameters change their values. The author of the paper [10] showed that the entropy of the Janus interface in a $2d \mathcal{N} = (2, 2)$ SCFT can be written by a specific linear combination of analytically continued Kähler potentials on moduli space:

$$S_{\mathcal{I}} \propto \left[K(\tau_+, \overline{\tau}_+) + K(\tau_-, \overline{\tau}_-) - K(\tau_+, \overline{\tau}_-) - K(\tau_-, \overline{\tau}_+) \right] =: \mathcal{D}.$$
(1.1)

Calabi named this quantity \mathcal{D} diastasis [11] which is an ancient Greek word meaning "distance". It reduces to a geodesic distance defined from the Zamolodchikov metric on the moduli space [11], but does not satisfy the axiom of distance, *e.g.* the triangle inequality [10]. The relation (1.1) in a 2*d* $\mathcal{N} = (2, 2)$ SCFT was further confirmed by holography in [12], super-Weyl anomaly in [13], and supersymmetric localization in [14].

Then a natural question is whether the relation [15] holds for other cases. It was conjectured [16] that the relation (1.1) also holds in $4d \mathcal{N} = 2$ SCFTs ³ from a holographic consideration. The main purpose of this thesis is to give a proof of this claim via supersymmetric localization based on our paper [15].

²The precise definition of the conformal manifold will be given in Section 2.3.1.

³The half-BPS superconformal Janus interface in a $4d \mathcal{N} = 2$ SCFT was studied in the previous works [17, 18, 19, 20, 21].

1.2 Summary of this thesis

Now let us summarize an outline of the proof [15] of the claim (1.1) in 4d $\mathcal{N} = 2$ SCFTs.

- We first define an interface entropy as a contribution to an entanglement entropy from an interface.
- We then derive the relation between an entropy of a conformal and superconformal interface and the partition function on a sphere by the conformal map introduced by [22].
- We evaluate the sphere partition function in the presence of the Janus interface in a $4d \mathcal{N} = 2$ SCFT via supersymmetric localization.
- Finally we combine this result and the above relation to show that the Janus interface entropy can be written by Calabi's diastasis.

The organization of this thesis is as follows. In Chapter 3 we define an interface entropy by using an entanglement entropy, and derive the relations between the interface entropy and partition functions in the presence of a conformal/superconformal interface. The results in Chapter 3 hold for general conformal/superconformal interfaces.

We focus on the Janus interface in $4d \mathcal{N} = 2$ SCFTs after Chapter 4. In Chapter 4 we give an off-shell construction of the Janus interface. In Chapter 5 we compute the sphere partition function via supersymmetric localization technique. By combining this result and the result in Chapter 3 we show that an interface entropy of the Janus interface in $4d \mathcal{N} = 2$ SCFTs can be written as Calabi's diastasis.

We give a brief review on CFT and SCFT in Chapter 2. Each chapter also has a brief review on relevant topics. Notations, conventions and technical issues are summarized in Appendix.

Chapter 2

Conformal and superconformal field theory

In this chapter we give a very brief review on conformal field theories (CFTs) and superconformal field theories (SCFTs). The main purpose is to explain symmetry and general structure of correlation functions and partition functions.

2.1 Conformal Field Theory

2.1.1 Conformal symmetry and conformal algebra

First let us explain about conformal field theories in general dimensions with Lorentzian signature ¹. We use a convention in which $\eta_{\mu\nu} = (-1, +1, \dots, +1)$. The conformal transformation is a transformation which preserves a metric up to a factor,

$$g'_{\mu\nu}(x') = \Omega(x)g_{\mu\nu}(x).$$
 (2.1)

Under infinitesimal coordinate transformations $x^{\mu} \to x^{\mu} + \epsilon^{\mu}(x)$, the metric transforms as

$$g_{\mu\nu} \to g_{\mu\nu} - \partial_{\mu}\epsilon_{\nu} - \partial_{\nu}\epsilon_{\mu}.$$
 (2.2)

Imposing the above metric transformation to be a conformal transformation (2.1), we have the following condition on ϵ :

$$\partial_{\mu}\epsilon_{\nu} + \partial_{\nu}\epsilon_{\mu} = \frac{2}{d}\partial_{\rho}\epsilon^{\rho}g_{\mu\nu}.$$
(2.3)

¹For more detail on CFTs, see for example [23, 24].

This equation is called the conformal Killing equation. Let us focus on the solution for $d \ge 3^2$. The general solutions for (2.3) are given by

$$\epsilon^{\mu} = a^{\mu} + \lambda^{\mu\nu} x_{\nu} + \lambda_D x^{\mu} + \left(x^2 \lambda_K^{\mu} - 2x^{\mu} x^{\nu} \lambda_{K\nu} \right), \qquad (2.4)$$

where $a^{\mu}, \lambda^{\mu\nu}, \lambda_D, \lambda_K^{\mu}$ are real parameters. We explain the transformations caused by each terms.

- The first term corresponds to translations $x'^{\mu} = x^{\mu} + a^{\mu}$ where a^{μ} are d real parameters.
- The second term corresponds to rotations $x'^{\mu} = \lambda^{\mu\nu} x_{\nu}$ where $\lambda^{\mu\nu}$ is real and antisymmetric tensor $\lambda^{\mu\nu} = -\lambda^{\nu\mu}$ which has $\frac{1}{2}d(d-1)$ real parameters.
- The third term corresponds to dilatations $x'^{\mu} = \lambda_D x^{\mu}$ where λ_D is a real parameter.
- The forth term corresponds to special conformal transformations $x'^{\mu} = x^2 \lambda_K^{\mu} 2x^{\mu} x^{\nu} \lambda_{K\nu}$ where λ_K^{μ} are *d* real parameters.

Thus in total the solutions (2.4) are parametrized by $d + \frac{1}{2}d(d-1) + 1 + d = \frac{1}{2}(d+1)(d+2)$ real parameters. We denote the corresponding generators by $P_{\mu}, M_{\mu\nu}, D, K_{\mu}$ to write conformal transformations as

$$\delta_C(\epsilon) = a^{\mu} + \frac{1}{2} \lambda^{\mu\nu} M_{\mu\nu} + \lambda_D D + \lambda_K^{\mu} K_{\mu}. \qquad (2.5)$$

The commutators of these generators are given by

$$[M_{\mu\nu}, M^{\rho\sigma}] = \eta_{\mu\rho} M_{\sigma\nu} - \eta_{\nu\rho} M_{\sigma\mu} - \eta_{\mu\sigma} M_{\rho\nu} + \eta_{\nu\sigma} M_{\rho\mu}, \qquad (2.6)$$

$$[P_{\mu}, M_{\nu\rho}] = 2\eta_{\mu[\nu}P_{\rho]}, \quad [K_{\mu}, M_{\nu\rho}] = 2\eta_{\mu[\nu}K_{\rho]}, \tag{2.7}$$

$$[P_{\mu}, K_{\nu}] = 2(\eta_{\mu\nu}D + M_{\mu\nu}), \qquad (2.8)$$

$$[D, P_{\mu}] = P_{\mu}, \quad [D, K_{\mu}] = K_{\mu}, \tag{2.9}$$

and the other commutators vanish³. The group generated by them are called the conformal group, which is equivalent to SO(d, 2). If we perform Wick rotation and consider the Euclidean signature, then the resulting conformal group is SO(d+1, 1).

 $^{^2\}mathrm{In}~d=2$ finite conformal symmetry is enhanced to infinite symmetry called Virasoro symmetry.

³We define $a_{[\mu}b_{\nu]} := \frac{1}{2}(a_{\mu}b_{\nu} - a_{\nu}b_{\mu}).$

2.1.2 Primary and descendant operator

Instead of demonstrating a whole story about a representation of conformal group in this thesis, we quickly summarize the conclusion. A representation is specified by an operator $\mathcal{O}(x)$ called a primary operator which satisfies the following conditions:

$$[D, \mathcal{O}(0)] = \Delta \mathcal{O}(0), \qquad (2.10)$$

$$[M_{\mu\nu}, \mathcal{O}(0)] = \mathcal{S}_{\mu\nu}\mathcal{O}(0), \qquad (2.11)$$

$$[K_{\mu}, \mathcal{O}(0)] = 0, \qquad (2.12)$$

where Δ is called a conformal dimension or a Weyl wight, and $S_{\mu\nu}$ is a representation matrix of Lorentz group satisfying the commutation relation (2.6). The other operator in this representation can be obtained by acting the operator P_{μ} on the primary operator. They are called descendants.

2.1.3 Correlation function in conformal field theory

Quantum field theories (QFTs) which are invariant under conformal transformations are called conformal field theories (CFTs). Correlation functions in CFTs are highly constrained by the conformal symmetry. As a result, *n*point functions with n = 1, 2, 3 of primary operators are completely fixed up to constant factors. Let us explain this point. First, all one point functions vanish on \mathbb{R}^d . Next, the two point function of real scalar primary operators \mathcal{O} with conformal dimension Δ on \mathbb{R}^d is determined as

$$\langle \mathcal{O}(x)\mathcal{O}(y)\rangle = \frac{k}{|x-y|^{2\Delta}},$$
(2.13)

where k is a constant which can be normalized as k = 1 in an unitary CFT. Furthermore, the three point function of primary operators \mathcal{O}_i with dimension Δ_i (i = 1, 2, 3) is given by

$$\langle \mathcal{O}_1(x_1)\mathcal{O}_2(x_2)\mathcal{O}_3(x_3) \rangle$$

= $\frac{c_{123}}{|x_1 - x_2|^{\Delta_1 + \Delta_2 - \Delta_3} |x_2 - x_3|^{\Delta_2 + \Delta_3 - \Delta_1} |x_3 - x_1|^{\Delta_3 + \Delta_1 - \Delta_2}},$ (2.14)

where c_{123} is a dynamical coefficient called an OPE (operator product expansion) coefficient. Higher points functions can be determined by two and

three point functions, *i.e.* spectrum and OPE coefficients in principle by using OPE. Thus these data are called CFT data 4 .

2.1.4 Trace anomaly

A CFT is a scale invariant theory by definition. This means that the trace of a stress-energy tensor $T_{\mu\nu}$ vanishes in classical levels:

$$\langle T_{\mu}{}^{\mu} \rangle = 0, \quad \text{(classical)}. \tag{2.15}$$

However, this does not hold in quantum levels for generic CFTs in even dimensions. This fact is called the trace anomaly or the Weyl anomaly or the conformal anomaly. They can be written as

$$\langle T_{\mu}{}^{\mu} \rangle = -2(-)^{d/2} A E_d + \sum_n B_n I_n,$$
 (2.16)

where I_n are the independent Weyl invariants with a Weyl weight -d, and E_d is the *d*-dimensional Euler density. The first and the other terms are called the A-type and the B-type anomaly, respectively. The coefficient A in two dimensions is proportional to the *c*-central charge and the coefficient in four dimensions is proportional to the *a*-central charge. They are very fundamental quantities in CFTs to characterize them partly because they have monotonic properties under RG flows in four and two dimensions, as explained in Introduction. They also appear the universal part of sphere partition functions as we will explain in Section 2.3.2 and an entanglement entropy as we will explain in Section 3.1.3. Note that in odd dimensions we do not have Weyl anomaly.

2.2 Supersymmetry and superconformal field theory

2.2.1 Supersymmetry

The supersymmetry (SUSY) is symmetry which exchanges bosons and fermions. It is an extension of the Poincaré group whose generators we denote by P_{μ}

⁴As we will explain later, in a CFT with an interface one point functions can have non-trivial values, so we have to include coefficients of one point functions to CFT data in this case.

2.2. SUSY AND SCFT

and $M_{\mu\nu}$. Representations of the SUSY algebra depend on spacetime dimensions. In this thesis we mostly focus on the 4d $\mathcal{N} = 2$ SUSY, so we here explain the SUSY in d = 4. We use the conventions and notations in [25] which are summarized in Appendix A. The SUSY transformations are generated by chiral generators $Q_{\alpha}^{i}, Q_{i\alpha}, (i = 1, \dots, \mathcal{N})$:

$$Q_i = P_L Q_i, \quad Q^i = P_R Q^i, \tag{2.17}$$

where $P_{L/R} = \frac{1}{2}(1\pm\gamma_*)$ are chiral projection operators. The Weyl conjugation of Q_i, Q^i which are denoted by $\overline{Q}_i, \overline{Q}^i$ are also chiral:

$$\overline{Q}_i = \overline{Q}_i P_L, \quad \overline{Q}^i = \overline{Q}^i P_R.$$
(2.18)

They satisfy the following commutation relations 5:

$$\{Q_{i\alpha}, \overline{Q}^{j\beta}\} = -\frac{1}{2}\delta_i^j (P_L \gamma_\mu)_\alpha{}^\beta P^\mu, \quad \{Q_\alpha^i, \overline{Q}_j^\beta\}, = -\frac{1}{2}\delta_j^i (P_R \gamma_\mu)_\alpha{}^\beta P^\mu, \quad (2.19)$$

$$\{Q_{i\alpha}, \overline{Q}_j^\beta\} = 0, \qquad \{Q_{\alpha}^i, \overline{Q}^{j\beta}\} = 0 \qquad (2.20)$$

$$[M_{[\mu\nu]}, Q_{i\alpha}], = -i\frac{1}{2}(\gamma_{\mu\nu})_{\alpha}{}^{\beta}Q_{i\beta}, \qquad [M_{[\mu\nu]}, Q_{\alpha}^{i}] = -i\frac{1}{2}(\gamma_{\mu\nu})_{\alpha}{}^{\beta}Q_{\beta}^{i}, \qquad (2.21)$$

$$[P_{\mu}, Q_{i\alpha}] = 0, \qquad [P_{\mu}, Q_{\alpha}^{i}] = 0. \qquad (2.22)$$

In terms of these generators, general SUSY transformations can be written as

$$\delta_Q(\epsilon) = \overline{\epsilon}^i Q_i + \overline{\epsilon}_i Q^i, \qquad (2.23)$$

where ϵ_i, ϵ^i are left/right handed SUSY parameters:

$$\epsilon_i = P_L \epsilon_i, \quad \epsilon^i = P_R \epsilon^i. \tag{2.24}$$

There is the symmetry which mixes \mathcal{N} -supercharges. This symmetry is called the *R*-symmetry.

2.2.2 Superconformal symmetry

A superconformal group contains conformal symmetry and R-symmetry as its bosonic subgroup. It also contains the symmetry generated by spinor generators Q^i, Q_i that are superpartners of translation generators P_{μ} and S^i, S_i that

 $^{^5\}mathrm{We}$ can have more general algebras by including the so-called central charge which we do not refer to.

are superpartners of special conformal generators K_{μ} . The former is called the Poincaré SUSY and the latter is called the special superconformal symmetry. Explicitly, the superconformal group in four dimensions is $SU(2, 2|\mathcal{N})$ for $\mathcal{N} = 1, 2, 3$ whose bosonic subgroup is $SO(4, 2) \times SU(\mathcal{N})_R \times U(1)_R$. The first factor is the conformal symmetry in four dimensions, and the rest factors are *R*-symmetry as indicated in subscripts. For $\mathcal{N} = 4$ case the superconformal group is PSU(2, 2|4).

The Poincaré SUSY transformations are generated by $\delta_Q(\epsilon)$ (2.23) while the special superconformal transformations are generated by

$$\delta_S(\eta) = \overline{\eta}^i S_i + \overline{\eta}_i S^i, \qquad (2.25)$$

where η^i, η_i are right/left handed SUSY parameters:

$$\eta^i = P_R \eta^i, \quad \eta_i = P_L \eta_i. \tag{2.26}$$

CFTs which admit superconformal symmetry are called superconformal field theories (SCFTs).

2.3 Sphere partition function

2.3.1 Conformal manifold

Consider a *d*-dimensional CFT and deform it by a set of operators $\{\mathcal{O}_I\}_{I=1}^n$. If we have a Lagrangian description this means that we add

$$\frac{1}{\pi^{d/2}} \int \mathrm{d}^d x \tau^I \mathcal{O}_I(x) \tag{2.27}$$

to an original Lagrangian where τ^{I} are parameters. If we do not have a Lagrangian description we insert

$$\exp\left[\frac{1}{\pi^{d/2}}\int \mathrm{d}^d x \tau^I \mathcal{O}_I(x)\right] \tag{2.28}$$

to correlation functions. If operators \mathcal{O}_I have a conformal dimension d, then this operators are called marginal operators and the corresponding deformation is called a marginal deformation. Furthermore, if the deformed theory is again conformal in all order in perturbation theory with respect to τ^I then the operators are called exactly marginal operators and the deformation is called an exactly marginal deformation. In this case, we have a family of CFTs parametrized by τ^{I} which we call the conformal manifold or the moduli space S. The parameters τ^{I} are called moduli parameters or exactly marginal couplings. It is known that the conformal manifold S admits the Riemannian metric called the Zamolodchikov metric [3].

2.3.2 Divergent structure of sphere partition function

In this subsection we explain the general structure of partition functions of CFTs on a *d*-dimensional sphere. Since a sphere has a finite volume, there are no IR divergences, but there are still UV divergences. So we introduce an UV cut-off Λ_{UV} to regularize the partition function. Then sphere partition function has the following structure for even d^{-6} :

$$\log Z[\mathbb{S}^d] = A_1(\tau) (r\Lambda_{UV})^{2n} + A_2(\tau) (r\Lambda_{UV})^{2n-2} + \dots + A_n(\tau) (r\Lambda_{UV})^2 + A(\tau) \log(r\Lambda_{UV}) + F_{2n}(\tau),$$
(2.29)

where d = 2n and r is a radius of the sphere. On the other hand, for odd d it has the following structure:

$$\log Z[\mathbb{S}^d] = B_1(\tau) (r\Lambda_{UV})^{2n+1} + B_2(\tau) (r\Lambda_{UV})^{2n-1} + \dots + B_n(\tau) (r\Lambda_{UV}) + F_{2n+1}(\tau).$$
(2.30)

The power law divergences correspond to the counterterms which are explicitly written as

$$\Lambda_{UV}^{2n-2k+2} \int d^{2n} x \sqrt{g} A_k(\tau) R^{k-1}$$
 (2.31)

for even d and

$$\Lambda_{UV}^{2n-2k+3} \int d^{2n+1}x \sqrt{g} B_k(\tau) R^{k-1}$$
(2.32)

for odd d. So these divergent terms are ambiguous and can be removed in the continuum limit.

For even d one can show that the coefficient in front of the log divergent term does not depend on the exactly marginal parameter τ and is proportional to the Weyl anomaly coefficient. On the other hand, the finite term

⁶Without loss of generality we consider a single moduli parameter τ and omit indices.

 $F_{2n}(\tau)$ is ambiguous and can be removed by local conterterms

$$\int d^{2n}x \sqrt{g} F_{2n}(\tau) E_{2n}.$$
(2.33)

As we will explain in the next subsection, when we treat a supersymmetric theory the situation can be different.

For odd d one can also show that the finite term $F_{2n+1}(\tau)$ is unambiguous and does not depend on the exactly marginal parameter τ .

To recap, the universal term in the log of the sphere partition function is the log divergent term for even d and the constant term for odd d.

2.3.3 Supsersymmetric partition function

When we consider SCFTs the structure of sphere partition functions can be different [26]. In this case we often regularize a partition function with a part of SUSY preserved. In the absence of supersymmetry, we consider diffeomorphism invariant counterterms. On the other hand, in SUSY preserving scheme we consider SUSY invariant and diffeomorphism invariant counterterms given by supergravity. Thus the allowed counterterms become more restricted. As a result, a finite term $F_{2n}(\tau)$ for even d which is ambiguous in non-SUSY cases can be unambiguous. Indeed, for a $2d \mathcal{N} = (2, 2)$ SCFT and a $4d \mathcal{N} = 2$ SCFT it gives Kähler potentials on moduli space [27, 28, 26, 29]⁷:

$$Z_{\text{SUSY}}[\mathbb{S}^2](\tau,\overline{\tau}) = e^{-K(\tau,\overline{\tau})}, \quad Z_{\text{SUSY}}[\mathbb{S}^4](\tau,\overline{\tau}) = e^{K(\tau,\overline{\tau})/12}.$$
 (2.34)

The 2*d* result was first conjectured in [27] and later proved by [28] via the SUSY localization technique ⁸ ⁹. The authors of [26] further gave a new proof of this claim by using supersymmetric Ward identity which does not depend on a Lagrangian description nor SUSY localization. The 4*d* result was first derived in [26] via SUSY localization and further proved by supersymmetric Ward identity in [29].

A choice of SUSY preserving regularization schemes corresponds to the choice of Kähler potentials on the conformal manifold. In other words, chang-

⁷For a 4d $\mathcal{N} = 1$ and a 2d $\mathcal{N} = (1, 1)$ SCFT, the finite terms are ambiguous and thus not physical quantity [26].

 $^{^{8}}$ SUSY localization will be reviewed in Section 5.1.

⁹Here we focus only on the constant term in (2.29). Thus these equalities hold up to moduli independent factors due to the Weyl anomaly.

2.3. SPHERE PARTITION FUNCTION

ing a regularization scheme causes the the Kähler transformations

$$K(\tau, \overline{\tau}) \to K(\tau, \overline{\tau}) + \mathcal{F}(\tau) + \overline{\mathcal{F}}(\overline{\tau}),$$
 (2.35)

where $\mathcal{F}(\cdot)$ and $\overline{\mathcal{F}}(\cdot)$ are a holomorphic and an anti-holomorphic function, respectively. This means that the sphere partition function is not actually a function but a section. This difference can be explained by supergravity counterterms [26, 29]. We will explain this point more explicitly in the presence of the Janus interface in Section 5.2.4.

Chapter 3

Interface entropy

In this chapter we study an entropy of a general conformal/superconformal interface. Although we explained motivations to study an interface entropy in Chapter 1, here we want to recap points. One motivation to study an interface entropy is that it is conjectured to be a *C*-function in a CFT with an interface (ICFT) [2]. If this statement is true, then an interface entropy can be used to classify/constraints data in ICFTs. Thus it is very important to compute an interface entropy nonperturbatively for examining its property. The second motivation that is more directly related to this thesis is to study the relation between an entropy of the Janus interface and Calabi's diastasis. This topic will be explained in Chapter 4 and Chapter 5.

Let us summarize contents covered in this chapter. The goal of this chapter is to derive the relation between a sphere partition function and an interface entropy in the presence of a conformal/superconformal interface which are given in (3.36) and (3.42). The main tool to derive these relations is a conformal map introduced in the paper [22] which we call the CHM map. So we first review in Section 3.1 the CHM map and the derivation of the relation between a sphere partition function and an entanglement entropy across a spherical entangling surface in the absence of an interface according to [22]. Next we introduce a conformal interface and define its entropy. Then we apply the CHM map to derive the similar relation in the presence of a conformal interface in Section 3.2. In Section 3.3 we further derive the similar relation for a half-BPS superconformal interface in a SCFT after defining it. In an intermediate step we make an assumption. We also discuss this assumption is quite natural in terms of the supersymmetric Rényi entropy.

3.1 Entanglement entropy and sphere partition function in CFT

In this section we review the results [22] in general CFTs without an interface to apply them to CFTs with an interface.

3.1.1 CHM map

First we explain the original map which transforms the Minkowski space $\mathbb{R}^{1,d-1}$ into the static patch of the de Sitter space. We parametrize the *d*-dimensional Minkowski space by (t, r, Ω_{d-2}) where Ω_{d-2} parametrize \mathbb{S}^{d-2} . The metric is given by

$$ds^{2} = -dt^{2} + dr^{2} + r^{2}d\Omega_{d-2}^{2}.$$
(3.1)

Next we consider the conformal transformation $(r, t, \Omega_{d-2}) \rightarrow (\theta, \tau, \Omega_{d-2})$

$$t = R \frac{\cos\theta \sinh(\tau/R)}{1 + \cos\theta \cosh(\tau/R)},$$
(3.2)

$$r = R \frac{\sin \theta}{1 + \cos \theta \cosh(\tau/R)},\tag{3.3}$$

where R is a constant and $0 \le \theta \le \pi/2$. We call this transformation as the (Lorentzian version of) CHM map [22]. Under this map, the metric (3.1) transforms as

$$ds^{2} = \Omega^{2} \left(-\cos^{2}\theta d\tau^{2} + R^{2} d\theta^{2} + \sin^{2}\theta d\Omega_{d-2}^{2} \right), \qquad (3.4)$$

where a conformal factor Ω is given by

$$\Omega = \frac{1}{1 + \cos\theta \cosh(\tau/R)}.$$
(3.5)

The new coordinates $(\theta, \tau, \Omega_{d-2})$ are the static patch of the de Sitter space with a radius R. This fact can be seen more transparently when we write the metric (3.4) by $(\hat{r} = R \sin \theta, \tau, \Omega_{d-2})$ as

$$ds^{2} = -\left(1 - \frac{\hat{r}^{2}}{R^{2}}\right)d\tau^{2} + \frac{d\hat{r}^{2}}{1 - \hat{r}^{2}/R^{2}} + \hat{r}^{2}d^{2}\Omega_{d-2}.$$
 (3.6)

3.1. ENTANGLEMENT ENTROPY IN CFT

The surface $\hat{r} = 1$ corresponds to a cosmological horizon.

On the other hand, if we start with the Euclidean flat space \mathbb{R}^d instead of Minkowski space in which the metric can be obtained by performing Wick rotation $t \to it$ in (3.1) as

$$ds^{2} = dt^{2} + dr^{2} + r^{2} d\Omega_{d-2}^{2}.$$
(3.7)

In this case, the (Euclidean version of) CHM map is given by

$$t = R \frac{\cos\theta \sin(\tau/R)}{1 + \cos\theta \cos(\tau/R)},$$
(3.8)

$$r = R \frac{\sin \theta}{1 + \cos \theta \cos(\tau/R)},\tag{3.9}$$

and the metric after this transformation is

$$\mathrm{d}s^2 = \Omega^2 \left(\cos^2\theta \mathrm{d}\tau^2 + R^2 \mathrm{d}\theta^2 + \sin^2\theta \mathrm{d}\Omega_{d-2}^2 \right) = \Omega^2 \mathrm{d}s_{\mathbb{S}^d}^2, \tag{3.10}$$

where a conformal factor Ω is given by

$$\Omega = \frac{1}{1 + \cos\theta \cos(\tau/R)},\tag{3.11}$$

and $0 \leq \theta \leq \pi/2$. So in this case we finally have a *d*-dimensional sphere \mathbb{S}^d .

3.1.2 Original derivation of the relation

By using the CHM map, the author of the paper [22] derived the relation between entanglement entropy with a spherical entangling surface and the sphere partition function. We review their results here to apply them to an ICFT later.

We start with the Lorentzian flat space $\mathbb{R}^{1,d-1}$, and take $\Sigma := \mathbb{S}^{d-2}$ on a constant time slice as an entangling surface. See Figure 3.1. We denote inside and outside of Σ as V and \tilde{V} respectively. Then by tracing out the degree of freedom in \tilde{V} we obtain a reduced density matrix ρ . An entanglement entropy S_E across Σ is given by the von Neumann entropy

$$S_E := -\operatorname{tr}_V \left[\rho \ln \rho\right], \qquad (3.12)$$

where a trace is taken for the degree of freedom in V. Note that an entanglement entropy in a continuum quantum field theory has UV divergent which



Figure 3.1: Entanglement entropy without an interface

we should regulate properly. We will explain this point in Section We will explain how to regularize UV divergent in Section 3.1.3.

Next, we use the CHM map (3.2). After this transformation, the modular flow induced by ρ^{is} inside the causal development of V is mapped to the time translation $\tau \to \tau + 2\pi Rs$ in de Sitter space ¹. Then one can show that states in the static de Sitter space is thermal at a temperature $T = 1/(2\pi R)$ and the density matrix is given by

$$\rho = \frac{e^{-\beta H_{\tau}}}{\operatorname{tr} e^{-\beta H_{\tau}}},\tag{3.13}$$

where H_{τ} is a operator generating the translation in τ -direction. Thus the entanglement entropy (3.12) can be written as

$$S_E = \beta E + \ln Z [\mathrm{dS}_d], \qquad (3.14)$$

where $\beta := 2\pi R$, $E := \operatorname{tr}(\rho H_{\tau})$, and $Z := \operatorname{tr} e^{-\beta H_{\tau}}$.

As explained in the next subsection, we are interested in the universal part of an entanglement entropy S_E which is a log divergent term for even d and a constant term for odd d. Since the energy term E is given by an integral of the one point function of the energy momentum tensor, it gives a finite result for even d and 0 for odd d, see Section 2.1.4. Therefore we do not need to consider this term for our purpose. Thus all we need is the

¹The causal development of V is a set of points such that all causal curves through them intersect V.

second term in (3.14) which becomes $\ln Z[\mathbb{S}^d]$ after Wick rotation. Therefore we can finally write the entanglement entropy across a spherical entangling surface by free energy $W := -\ln Z[\mathbb{S}^d]$ as

$$S_E = -W = \ln Z[\mathbb{S}^d]. \tag{3.15}$$

3.1.3 UV divergence

Let us mention about UV divergence. In general, an entanglement entropy in a continuum QFT has the following UV structure:

$$S_E = (\text{power low divergences}) + \begin{cases} a_d \log(r\Lambda_{UV}) + f_d & (d:\text{even}) \\ f_d & (d:\text{odd}) \end{cases}.$$
 (3.16)

The log divergent term in even dimensions and the constant term in odd dimensions are known to be universal, i.e. independent of UV regularization schemes.

On the other hand, the log of the sphere partition function also has the same divergence structure as (2.29) and (2.30). Then the precise version of the equation (3.15) is that

$$S_E|_{\log} = \ln Z[\mathbb{S}^d]|_{\log}, \qquad (d: \text{even}), \qquad (3.17)$$

$$S_E|_{\text{const}} = \ln Z[\mathbb{S}^d]|_{\text{const}}, \qquad (d: \text{odd}). \qquad (3.18)$$

3.1.4 Another derivation

In the previous section, we derive the relation (3.15) between the free energy and entanglement entropy. Here we introduce another equivalent derivation via replica methods according to [2]. Here we assume that we are in even dimensions and using dimensional regularization to avoid an extra care about the conformal anomaly in intermediate steps. It is automatically incorporated into a pole in even dimensions.

We first introduce the quantity called the Rényi Entropy S_n which is parametrized by a real parameter n called the Rény parameter as

$$S_n := \frac{1}{1-n} \ln \operatorname{tr}_V \rho^n.$$
(3.19)

This quantity reduces to an entanglement entropy (3.12) by taking $n \to 1$ limit:

$$\lim_{n \to 1} S_n = S_E. \tag{3.20}$$

In the replica trick we identify the Rényi entropy S_n with the partition function on a branched cover of the original manifold \mathcal{M} which we denote by \mathcal{M}_n^2 . With a proper normalization, this can be stated as

$$\operatorname{tr}_{V}\rho^{n} = \frac{Z[\mathcal{M}_{n}]}{Z[\mathcal{M}]^{n}}.$$
(3.21)

In this case $\mathcal{M} = \mathbb{R}^d$. We use the Euclidean version of the CHM map (3.8). The metric is conformally equivalent to the one of sphere, $ds_{\mathbb{R}^d}^2 = \Omega^2 ds_{\mathbb{S}^d}^2$ via the CHM map. Moreover, the *n*-fold cover of \mathbb{R}^d is mapped to \mathbb{S}_n^d as

$$\mathrm{d}s_{\mathbb{R}_n^d}^2 = \Omega^2 \mathrm{d}s_{\mathbb{S}_n^d}^2,\tag{3.22}$$

where

$$\mathrm{d}s_{\mathbb{S}_n^d}^2 = \cos^2\theta \mathrm{d}\tau^2 + R^2 n^2 \mathrm{d}\theta^2 + \sin^2\theta \mathrm{d}\Omega_{d-2}^2. \tag{3.23}$$

So the partition functions on each manifolds are related as

$$Z[\mathbb{R}_n^d] = Z[\mathbb{S}_n^d], \quad Z[\mathbb{R}^d] = Z[\mathbb{S}^d]$$
(3.24)

Note that we do not need care about the conformal anomaly in this step since we use the dimensional regularization.

Thus we can obtain the Rényi entropy via

$$S_n = \frac{1}{1-n} \ln \operatorname{tr}_V \rho^n = \ln \frac{Z[\mathbb{S}_n^d]}{Z[\mathbb{S}^d]^n}.$$
 (3.25)

To obtain the entanglement entropy, we evaluate the above equation around n = 1.

$$S_E = \lim_{n \to 1} \frac{1}{1 - n} \left(\ln Z[\mathbb{S}_n^d] - \ln Z[\mathbb{S}^d] \right) + \ln Z[\mathbb{S}^d]$$
(3.26)

The first term around n = 1 can be written as

$$\ln Z[\mathbb{S}_n^d] - \ln Z[\mathbb{S}^d] = -\frac{1}{2} \int_{\mathbb{S}^d} \delta g_{\mu\nu} \left\langle T^{\mu\nu} \right\rangle_{\mathbb{S}^d} + \mathcal{O}\left((n-1)^2 \right), \qquad (3.27)$$

²More precisely \mathcal{M}_n is made as follows. First we prepare *n*-th copies of \mathcal{M} . Then we cut and glue them at entangling surfaces.

where $\delta g_{\mu\nu}$ is a deviation of the \mathbb{S}_n^4 metric from \mathbb{S}^4 up to O(n-1). In the absence of the conformal anomaly, the one point function of the energy momentum tensor on sphere is proportional to that on flat space as

$$\langle T^{\mu\nu} \rangle_{\mathbb{S}^d} = (\text{Weyl factor})^2 \langle T^{\mu\nu} \rangle_{\mathbb{R}^d} = 0.$$
 (3.28)

So we show that the first term in (3.26) vanishes to have $S_E = \ln Z[\mathbb{S}^d]$. This is exactly the same result in the previous subsection.

3.2 Entropy of a conformal interface

In this section we define a conformal interface and its entropy. Then we derive the relation between a sphere partition function and an interface entropy by generalizing the results reviewed in the previous section.

3.2.1 Conformal interface

We consider non-local objects in QFTs that are called defects. Let us consider a CFT in *d*-dimensional flat space \mathbb{R}^d and insert a co-dimesion *p* defect. We are interested in defects which preserve a part of conformal symmetry SO(d+1,1). We call defects which preserve the maximal subgroup $SO(p+1,1) \times SO(q)$ as conformal defects and CFTs with defects as defect conformal field theories (DCFTs). In particular co-dimension one defects are denoted as interfaces, domain-walls, and boundaries. An interface preserving $SO(d,1) \subset$ SO(d+1,1) is called as a conformal interface and a CFT with a conformal interface as an interface CFT (ICFT).

Although the full conformal invariance is violated by inserted defects, the correlation functions in DCFTs or ICFTs are strongly constrained by the remaining conformal symmetry. For example, one can show that one point functions in an ICFT of an operator with nonzero spin vanish [30]. On the other hand, the one point function of scalar operators \mathcal{O} with a conformal dimension $\Delta_{\mathcal{O}}$ is given by

$$\left\langle \mathcal{O}(y) \right\rangle^{(\text{ICFT})} = \frac{a_{\mathcal{O}}}{\left| y_{\perp} \right|^{\Delta_{\mathcal{O}}}},$$
(3.29)

where we introduce Cartesian coordinate y^{μ} and y_{\perp} is a coordinate perpendicular to the interface. The nontrivial coefficient $a_{\mathcal{O}}$ which vanish in a CFT without an interface is also dynamical data in an ICFT.



Figure 3.2: Entanglement entropy with an interface

3.2.2 Entropy of conformal interface

Let us explain setup depicted in Figure 3.2. We start with a flat space \mathbb{R}^d and parametrize it by Cartesian coordinates $(y^0, y^1, \dots, y^{d-1})^3$. A planar interface \mathcal{I} is placed at $y^{d-1} = 0$. This preserves SO(d, 1) and therefore is a conformal interface. Two different CFTs live in both sides of the interface, and we denote it by CFT₊ and CFT₋, respectively.

Then we define an entropy associated to this interface. Frist we take $\Sigma := \mathbb{S}^{d-2}$ as an entangling surface. Then we have an entanglement entropy in the presence of an interface by (3.12) which we denote as $S_E^{(\text{ICFT})}$. We also have an entanglement entropy across the same surface Σ in the absence of an interface for a CFT₊ and CFT₋. We denote them by $S_E^{(\text{CFT}\pm)}$. Then a definition of an interface entropy $S_{\mathcal{I}}$ is given as a contribution to an entanglement entropy from an interface:

$$S_{\mathcal{I}} := S_E^{(\text{ICFT})} - \frac{1}{2} \left(S_E^{(\text{CFT}_+)} + S_E^{(\text{CFT}_-)} \right).$$
(3.30)

In the absence of an interface we can use the result reviewed in Section 3.1:

$$S_E^{(\mathrm{CFT}_{\pm})} = \ln Z^{(\mathrm{CFT}_{\pm})}[\mathbb{S}^d].$$
(3.31)

³From now on we denote Cartesian coordinates in a flat space by y^{μ} to distinguish them from stereographic coordinates on a sphere denoted by x^{μ} which will be introduced in the next chapter.

In the presence of an interface, we again use the CHM map to map \mathbb{R}^d to \mathbb{S}^d after the Wick rotation. By this map, a planar interface is mapped to spherical surface at an equator of \mathbb{S}^d . Similar as in a CFT without an interface, an entanglement entropy with an interface can be written as

$$S_E^{(\text{ICFT})} = \lim_{n \to 1} \frac{1}{1 - n} \left(\ln Z^{(\text{ICFT})}[\mathbb{S}_n^d] - \ln Z^{(\text{ICFT})}[\mathbb{S}^d] \right) + \ln Z^{(\text{ICFT})}[\mathbb{S}^d] \quad (3.32)$$

via the CHM map. The first term around n = 1 can be written as

$$\ln Z^{(\text{ICFT})}[\mathbb{S}_n^d] - \ln Z^{(\text{ICFT})}[\mathbb{S}^d] = -\frac{1}{2} \int_{\mathbb{S}^d} \delta g_{\mu\nu} \left\langle T^{\mu\nu} \right\rangle_{\mathbb{S}^d}^{(\text{ICFT})} + \mathcal{O}\left((n-1)^2 \right).$$
(3.33)

As we explained in 3.2.1, one point functions of operators with nonzero spins vanish due to the remaining conformal symmetry. So we can drop the first term since the one point function on a sphere is proportional to it on a flat space:

$$\langle T^{\mu\nu} \rangle_{\mathbb{S}^d}^{(\text{ICFT})} = (\text{Weyl factor})^2 \langle T^{\mu\nu} \rangle_{\mathbb{R}^d}^{(\text{ICFT})} = 0.$$
 (3.34)

So we can again show that the first term in (3.32) vanishes and thus

$$S_E^{(\text{ICFT})} = \ln Z^{(\text{ICFT})}[\mathbb{S}^d].$$
(3.35)

Then by combining the relations (3.31) and (3.31) with the definition of an interface entropy (3.30) we have ⁴

$$S_{\mathcal{I}} = \ln \left[\frac{Z^{(\text{ICFT})}[\mathbb{S}^d]}{(Z^{(\text{CFT}_+)}[\mathbb{S}^d]Z^{(\text{CFT}_-)}[\mathbb{S}^d])^{1/2}} \right].$$
(3.36)

3.2.3 UV divergence

Let us here mention about UV divergence in the interface entropy. As explained in Section 3.1.3, each terms in (3.30) have UV divergence. We focus on the even d case which will be used later. In this case, the entanglement entropy has the structure given in (3.16). Thus the interface entropy has the

 $^{^{4}}$ This result is the special case of the result in [2] where they considered more general conformal defects.

structure as 5

$$S_{\mathcal{I}} = (\text{power low divergence}) + f_d^{\mathcal{I}}.$$
 (3.37)

We explained that the constant terms in an entanglement entropy are ambiguous and thus not physical. However, their difference $f_d^{\mathcal{I}}$ is considered to be universal ⁶. So the formula (3.36) should be understood as the equality between finite contributions.

3.3 Entropy of a half-BPS superconformal interface

In this section, we define a half-BPS superconformal interface in a SCFT and derive the similar formula as (3.36) for a superconformal interface. We focus on CFTs with specific dimensions and SUSY, $4d \mathcal{N} = 2$ SCFTs.

3.3.1 Half-BPS superconformal interface

Let us consider a $\mathcal{N} = 2$ superconformal field theory on a flat space. The superconformal algebra is SU(2,2|2) whose bosonic subgroup is $SO(4,2) \times SU(2)_R \times U(1)_R$ as explained in Section 2.2.2. We are interested in an interface which maximally preserves a part of SU(2,2|2) symmetry. Such an interface is called a half-BPS superconformal interface.

As explained in Section 2.2.2, the full symmetry SU(2,2|2) is generated by Q^i, Q_i and S^i, S_i as

$$\delta_Q = \overline{\epsilon}^i Q_i + \overline{\epsilon}_i Q^i, \quad \delta_S = \overline{\eta}^i S_i + \overline{\eta}_i S^i. \tag{3.38}$$

A half-BPS superconformal interface is invariant under transformations by parameters satisfying

$$\epsilon_i = \rho_{ij} \gamma^3 \epsilon^j, \quad \eta_i = -\rho_{ij} \gamma^3 \eta^j, \tag{3.39}$$

30

⁵The log divergence has subtleties. This term arises from the conformal anomaly. The conformal anomaly in the presence of an interface has the contribution from bulk and the interface. One can show that the bulk contribution cancels out in (3.36), while we do not have a proof for vanishing of the contribution from the interface. However, holographic examples in [2] and Section 5.4 provide non-trivial evidences.

⁶For example the author of the paper [31] showed that this part is independent of the choice of UV regularization schemes by explicit calculation. This fact can be seen also in Section 5.4.

where ρ_{ij} satisfies $\rho_{ij} = \rho_{ji}$, $\rho_{ij}\overline{\rho}^{jk} = \delta_i^k$ with $\overline{\rho}^{ij} := (\rho_{ij})^*$. In other words, the preserved symmetry is generated by the following specific linear combination of supercharges:

$$Q_i - \rho_{ij}\gamma_3 Q^j, \quad S_i + \rho_{ij}\gamma_3 S^j. \tag{3.40}$$

The resulting symmetry is the $3d \mathcal{N} = 2$ superconformal algebra denoted as $OSp(2|4)_{sc}$ ⁷.

3.3.2 Entropy of a supefconformal interface

Now we want to derive similar relation as (3.36) for a half-BPS superconformal interface \mathcal{I} .

In this case, there are deferences from the previous argument. First, fields in SCFTs can couple to fields in a supergravity multiplet other than a metric. This makes a partition function of a SCFT a functional of a supergravity multiplet rather than a functional of a metric.

Second, we use a regularization scheme preserving the conformal symmetry in the previous derivation. In supersymmetric case, we have another choice; a regularization scheme which preserve a part of supersymmetry ⁸. We denote a partition function on a spacetime manifold \mathcal{M} in a SCFT in the presence of a half-BPS superconformal interface \mathcal{I} as $Z_{SUSY}^{\mathcal{I}}[\mathcal{M}]$. As we will see later in Chapter 5 it can be complex in general. If we simply replace $Z^{(ICFT)}[\mathbb{S}^d]$ in (3.36) with $Z_{SUSY}^{\mathcal{I}}[\mathbb{S}^d]$, the resulting entropy becomes complex. So we here make the following assumption:

$$Z^{(\text{ICFT})}[\mathbb{S}^4] = \left| Z^{\mathcal{I}}_{\text{SUSY}}[\mathbb{S}^4] \right|.$$
(3.41)

Then we obtain a relation between a SUSY partition function and an interface entropy for a half-BPS interface \mathcal{I} as

$$S_{\mathcal{I}} = \log \left[\frac{\left| Z_{\text{SUSY}}^{\mathcal{I}} [\mathbb{S}^4] \right|}{\left(Z_{\text{SUSY}}^{(\text{CFT}_+)} [\mathbb{S}^4] Z_{\text{SUSY}}^{(\text{CFT}_-)} [\mathbb{S}^4] \right)^{1/2}} \right] , \qquad (3.42)$$

where $Z_{SUSY}^{(CFT_{\pm})}[\mathbb{S}^4]$ is a partition function for CFT_{\pm} computed in a SUSY

⁷We write the letter "sc" to distinguish the $3d \mathcal{N} = 2$ superconformal algebra from $4d \mathcal{N} = 2$ massive SUSY algebra which we will denote $OSp(2|4)_{\rm m}$. We will explain their correspondence later.

⁸One example is SUSY localization which is reviewed in Section 5.1.

preserving scheme. Unfortunately we do not have a proof for the assumption (3.41), but we have some evidences for it ⁹.

- The assumption (3.41) is equivalent to an assumption that an entanglement entropy with an interface $S_E^{(\text{ICFT})}$ can be derived from the supersymmetric Rényi entropy. See the next subsection.
- The results obtained in holographic calculation are consistent with the formula (3.42). See Section 5.4.
- In two dimensional case, we can give a proof of similar relation (C.8) by using boundary super-Weyl anomaly. See Appendix C.

3.3.3 Supersymmetric Rényi entropy

Now we want to explain that the assumption (3.41) is natural in terms of the supersymmetric Rényi entropy. First let us define the supersymmetric Rényi entropy [33]. As we explained before, fields in SCFTs can couple to the supergravity background. Similar as the Rényi entropy, we consider the *n*-fold cover of a supergravity background on \mathbb{S}^4 which has singularity at n = 1. To define a supersymmetric Rényi entropy, we resolve this singularity and denote the resulting background as $\tilde{\mathbb{S}}_n^4$. The metric of the resolved *n*-fold cover $\tilde{\mathbb{S}}_n^4$ is given by

$$\mathrm{d}s_{\tilde{\mathbb{S}}_n^d}^2 = \left(f(\theta)\right)^2 \mathrm{d}\theta^2 + n^2 \sin^2\theta \mathrm{d}\tau^2 + \cos^2\theta \mathrm{d}\Omega_2^2,\tag{3.43}$$

where we set R = 1 and a smooth function $f(\theta)$ satisfies

$$f(\theta \to 0) = n, \quad f(\theta \gg \delta) = 1,$$
 (3.44)

for a small parameter $\delta \ll 1$. Then the supersymmetric Rényi entropy in the presence of a half-BPS interface \mathcal{I} denoted by $S_{\text{SUSY }n}^{\mathcal{I}}$ is given by

$$S_{\text{SUSY }n}^{\mathcal{I}} := \frac{1}{1-n} \operatorname{Re} \log \frac{Z_{\text{SUSY}}^{\mathcal{I}}[\widetilde{\mathbb{S}}_{n}^{4}]}{\left(Z_{\text{SUSY}}^{\mathcal{I}}[\mathbb{S}^{4}]\right)^{n}} .$$
(3.45)

⁹In d = 3, it is usual to define the free energy by taking the absolute value of the partition function as $F = -\log |Z_{SUSY}[\mathbb{S}^3]|$. See for example [6, 32].

Now we assume that an entanglement entropy with an interface can be derived from the supersymmetric Rényi entropy:

$$S_E^{(\text{ICFT})} = \lim_{n \to 1} S_{\text{SUSY } n}^{\mathcal{I}} .$$
(3.46)

Below we want to show that the formula (3.42) can be derived from this assumption and therefore the assumption (3.46) is equivalent to the assumption (3.41).

By substituting the equation (3.45) to the definition of an interface entropy (3.30), we have

$$S_E^{(\text{ICFT})} = \log \left| Z_{\text{SUSY}}^{\mathcal{I}} [\mathbb{S}^4] \right| - \partial_n \operatorname{Re} \log \left| Z_{\text{SUSY}}^{\mathcal{I}} [\widetilde{\mathbb{S}}_n^4] \right|_{n=1} .$$
(3.47)

Let us proof that the second term in (3.47) vanish. The background Weyl mutiplet

$$\mathcal{W} = (g_{\mu\nu}, \psi_{\mu}{}^{i}, V_{\mu}{}^{j}, A_{\mu}, D, \chi^{i}, T^{\pm}_{\mu\nu}) . \qquad (3.48)$$

couples to the supercurrent multiplet \mathcal{J} which contains a stress-energy tensor ¹⁰:

$$\mathcal{J} = (T_{\mu\nu}, S^i_{\mu}, j^{ij}_{\mu}, j_{\mu}, J, j^i, j^{\pm}_{\mu\nu}) .$$
(3.49)

Then the partition function $Z_{\text{SUSY}}^{\mathcal{I}}[\widetilde{\mathbb{S}}_n^4]$ on the branched sphere is expanded around n = 1 as

$$-\operatorname{Re} \log Z_{\mathrm{SUSY}}^{\mathcal{I}}[\widetilde{\mathbb{S}}_{n}^{4}] + \operatorname{Re} \log Z_{\mathrm{SUSY}}^{\mathcal{I}}[\mathbb{S}^{4}]$$

$$= \int_{\mathbb{S}^{4}} \mathrm{d}^{4}x \sqrt{g} \left\langle \frac{1}{2} \,\delta g_{\mu\nu} T^{\mu\nu} + \delta \overline{\psi}_{\mu}{}^{i} S^{\mu}{}_{i} + \delta \overline{\psi}_{\mu i} S^{\mu i} + \delta V_{\mu}{}^{ij} j^{\mu}{}_{ij} + \delta A_{\mu} j^{\mu} \right.$$

$$\left. + \delta D J + \delta \overline{\chi}{}^{i} j_{i} + \delta \overline{\chi}{}_{i} j^{i} + \delta T^{+\mu\nu} j_{+\mu\nu} + \delta T^{-\mu\nu} j_{-\mu\nu} \right\rangle_{\mathbb{S}^{4}}^{(\mathrm{ICFT})} + \mathcal{O}((n-1)^{2})$$

$$(3.50)$$

As in Section 3.2 contributions from operators with nonzero spins vanish. Thus we look into the one point function $\langle J \rangle_{\mathbb{S}^4}^{(\text{ICFT})}$. For this purpose, let us consider the SUSY transformation of j_i in the flat space:

$$\delta j_i = -\frac{1}{2} (\partial J) \epsilon_i + \frac{1}{2} j_{\mu i}{}^j \gamma^\mu \epsilon_j + \frac{i}{2} j_\mu \gamma^\mu \epsilon_i + j^-_{\mu\nu} \gamma^{\mu\nu} \varepsilon_{ij} \epsilon^j, \qquad (3.51)$$

 $^{^{10}}$ The explicit form of the supercurrent multiplet for a vector multiplet is given in Appendix A.6.

where constant SUSY parameters ϵ^i , ϵ_i satisfy the half-BPS condition (3.39). Then we take vacuum expectation values of both sides. Again one point functions of operators with nonzero spins vanish. We also know that $\langle \delta j_i \rangle_{\mathbb{R}^4}^{(\text{ICFT})} = 0$ from the Ward identities. Thus for constant ϵ^i , ϵ_i satisfying the condition (3.39) we have

$$0 = \partial_{\mu} \langle J \rangle_{\mathbb{R}^4}^{(\text{ICFT})} \gamma^{\mu} \epsilon_i . \qquad (3.52)$$

On the other hand, we know that the operator J has the Weyl weight 2 and thus 11

$$\langle J(y) \rangle_{\mathbb{R}^4}^{(\text{ICFT})} = \frac{a_J}{|y^3|^2},\tag{3.53}$$

where a_J is a constant. Substituting this equation into (3.52), we show that $a_J = 0$. So we conclude that $\langle J \rangle_{\mathbb{S}^4}^{(\text{ICFT})} = 0$ by the similar argument as (3.34).

Thus we show that the second term in (3.47) vanish and therefore

$$S_E^{(\text{ICFT})} = \log \left| Z_{\text{SUSY}}^{\mathcal{I}} [\mathbb{S}^4] \right| \quad . \tag{3.54}$$

By comparing this result and (3.35) we reach the assumption (3.41). Conversely, if we assume (3.41) we obtain (3.54) which is same as the supersymmetric Rényi entropy (3.45), thus we arrive at the assumption (3.46). So we conclude that the assumption (3.45) isequivalent to the assumption (3.41).

 $^{^{11}\}mathrm{Remind}$ that this form is determined by the remaining conformal symmetry. See Section 3.2.1.
Chapter 4

Construction of the Janus interface

So far we have considered general conformal/superconformal interfaces. From now on, we consider a specific interface called the Janus interface. The goal of the latter half is to show the relation between the Janus interface in 4d $\mathcal{N} = 2$ SCFTs and a specific linear combination of analytically continued Kähler potentials on the moduli space called Calabi's diastasis.

In this Chapter we explain a construction of the Janus interface. In Section 4.1, we explain an off-shell construction of the Janus interface in the flat space. We introduce a crucial ingredient, the coupling multiplet whose bottom component is a position dependent coupling $\tau(x)$. We will check the resulting interface is indeed a half-BPS superconformal interface. In Section 4.2 we construct the Janus interface on the four sphere and discuss the relation to the Janus interface in the flat space. In Section 4.3, we explicitly construct the Janus interface in a $4d \mathcal{N} = 2$ gauge theory which will be used in the computation in the next chapter.

4.1 Off-shell Construction of Janus interface in flat space

First we explain an off-shell construction of the Janus interface in flat space ¹. We denote Cartesian coordinates in flat space as y^{μ} , ($\mu = 0, 1, 2, 3$) and place

¹The similar method was used in [34, 35, 36, 14, 37] for various defects.

the Janus interface orthogonal to the y^3 -direction, see Figure 3.2. As we explained in Introduction, the Janus interface is an interface across which an exactly marginal coupling τ changes its values. To construct such an object we promote a coupling constant τ to a position-depending coupling field $\tau(y)$. Further, to construct a half-BPS interface we promote the coupling field $\tau(y)$ to the bottom component of a $\mathcal{N} = 2$ chiral multiplet

$$\mathcal{T} = (\tau, \Psi_i^{(\tau)}, B_{ij}^{(\tau)}, F_{\mu\nu}^{(\tau)-}, \Lambda_i^{(\tau)}, C^{(\tau)}).$$
(4.1)

We denote this multiplet by a coupling multiplet. As we will see soon later, by solving symmetry preserving conditions we can write all components in \mathcal{T} in terms of $\tau(y)$. We also have the anti-chiral counterpart of \mathcal{T} which we denote as $\overline{\mathcal{T}}$. The components of $\overline{\mathcal{T}}$ are given by

$$\overline{\mathcal{T}} = (\overline{\tau}, \Psi^{(\tau)i}, B^{(\tau)ij}, F^{(\tau)+}_{\mu\nu}, \Lambda^{(\tau)i}, \overline{C}^{(\tau)}), \qquad (4.2)$$

where we take $\overline{\tau}$ to be the complex conjugate of τ : $\overline{\tau} = \tau^*$.

Now we consider symmetry preserving conditions on \mathcal{T} . First, to preserve a part of Lorentz symmetry we set $F_{\mu\nu}^{(\tau)\pm} = 0$. Next, to preserve the half of supersymmetry, we impose that fermionic components in $\mathcal{T}, \overline{\mathcal{T}}$ and their SUSY variation vanish. Using the SUSY transformations of a chiral multiplet ², the conditions are written as

$$\delta \Psi_i^{(\tau)} = (\partial_3 \tau) \gamma^3 \epsilon_i + \frac{1}{2} B_{ij}^{(\tau)} \epsilon^j = 0 , \qquad (4.3)$$

$$\delta \Lambda_i^{(\tau)} = -\frac{1}{2} \partial_3 B_{ij}^{(\tau)} \varepsilon^{jk} \gamma^3 \epsilon_k + \frac{1}{2} C^{(\tau)} \varepsilon_{ij} \epsilon^j = 0 , \qquad (4.4)$$

$$\delta \Psi^{(\tau)i} = (\partial_3 \overline{\tau}) \gamma^3 \epsilon^i + \frac{1}{2} B^{(\tau)ij} \epsilon_j = 0 , \qquad (4.5)$$

$$\delta \Lambda^{(\tau)i} = -\frac{1}{2} \partial_3 B^{(\tau)ij} \varepsilon_{jk} \gamma^3 \epsilon^k + \frac{1}{2} \overline{C}^{(\tau)} \varepsilon^{ij} \epsilon_j = 0 , \qquad (4.6)$$

where ϵ^i, ϵ_i are constant SUSY parameters. By solving these equations, we have

$$B_{ij}^{(\tau)} = -2\rho_{ij}\,\partial_3\tau \,, \qquad C^{(\tau)} = -2e^{+2i\alpha}\,\partial_3^2\tau \,, \qquad (4.7)$$

$$B^{(\tau)ij} = -2\overline{\rho}^{ij}\,\partial_3\overline{\tau} \qquad \overline{C}^{(\tau)} = -2e^{-2i\alpha}\,\partial_3^2\overline{\tau} , \qquad (4.8)$$

²See appendix A.3.2.

4.1. CONSTRUCTION IN FLAT SPACE

for fields in the coupling multiplet and

$$\epsilon_i = \rho_{ij} \gamma^3 \epsilon^j, \tag{4.9}$$

for SUSY parameters. These solutions are parametrized by matrices $\rho_{ij}, \overline{\rho}^{ij}$ given as

$$\rho_{ij} = e^{i\alpha} \, \vec{n} \cdot \vec{\tau}_{ij} \,, \qquad \overline{\rho}^{ij} = e^{-i\alpha} \, \vec{n} \cdot \vec{\tau}^{ij} \,, \qquad (4.10)$$

where α is a real parameter and \vec{n} is a three-dimensional unit vector. The condition on SUSY parameters is equivalent to the first condition in (3.39). So we construct an interface which preserves the half of the supersymmetry. We further confirm that this interface is also invariant under special superconformal transformations generated by SUSY parameters satisfying the second condition in (3.39) when taking a step function profile in the next subsection.

Note that the auxiliary fields satisfy reality conditions even for generic $\tau(y)$:

$$B^{(\tau)ij} = (B_{ij}^{(\tau)})^*, \quad \overline{C}^{(\tau)} = (C^{(\tau)})^*.$$
(4.11)

This situation is different from the Janus interface on \mathbb{S}^4 which we will construct in Section 4.2.

4.1.1 Step function profile

So far we have considered an interface for a general profile $\tau(y)$. We now consider a step function profile:

$$\tau(y^3) = \begin{cases} \tau_+ & \text{for } y^3 > 0 ,\\ \tau_- & \text{for } y^3 < 0 . \end{cases}$$
(4.12)

In this case, the derivatives of $\tau(x)$ are given by a delta function as

$$\partial_3 \tau = \Delta \tau \,\delta(y^3), \quad \partial_3^2 \tau = \Delta \tau \,\delta'(y^3).$$
 (4.13)

So by substituting these expressions into (4.7) and (4.8), we can obtain the auxiliary fields for a step function profile as

$$B_{ij}^{(\tau)} = -2\rho_{ij}\Delta\tau\,\delta(y^3)\,,\quad C^{(\tau)} = -2e^{+2i\alpha}\Delta\tau\,\delta'(y^3)\,,$$

$$B^{(\tau)ij} = -2\overline{\rho}^{ij}\Delta\overline{\tau}\,\delta(y^3)\,,\quad \overline{C}^{(\tau)} = -2e^{-2i\alpha}\Delta\overline{\tau}\,\delta'(y^3)\,,$$
(4.14)

where $\Delta \tau := \tau_+ - \tau_-$.

Now we show that the Janus interface for a step function profile is invariant under special superconformal transformations. The special superconformal transformations δ_{η} are obtained by substitute $\epsilon^i \to y^{\mu} \gamma_{\mu} \eta^i$ in transformation rules in Appendix A.3.1. Then the special superconformal transformation of $\Psi_i^{(\tau)}$ automatically vanishes: $\delta_{\eta} \Psi_i^{(\tau)} = 0$. On the other hand, the special superconformal transformation of $\Lambda_i^{(\tau)}$ is given by

$$\delta_{\eta} \Lambda_{i}^{(\tau)} = -\frac{1}{2} \partial_{3} B_{ij}^{(\tau)} \varepsilon^{jk} y^{\mu} \gamma^{3} \gamma_{\mu} \eta_{k} + \frac{1}{2} C^{(\tau)} \varepsilon_{ij} y^{\mu} \gamma_{\mu} \eta^{j} - B_{ij}^{(\tau)} \varepsilon^{jk} \eta_{k} \qquad (4.15)$$
$$= 2\Delta \tau \, \partial_{3} (y^{3} \delta(y^{3})) \rho_{ij} \varepsilon^{jk} \eta_{k}$$

$$-\Delta\tau e^{2i\alpha} \,\delta'(y^3) \Big(y^3 + \sum_{a=0}^2 y^a \gamma_a \gamma^3 \Big) \varepsilon_{ij} \overline{\rho}^{jk} (\eta_k + \rho_{kl} \gamma^3 \eta^l) \,. \quad (4.16)$$

The first term in the final expression vanishes as a distribution, while the second term vanishes for SUSY parameters satisfying the condition (3.39). Thus we conclude that $\delta_{\eta} \Psi_i^{(\tau)} = \delta_{\eta} \Lambda_i^{(\tau)} = 0$ for a step function profile. Similarly, we can show that $\delta_{\eta} \Psi^{(\tau)i} = \delta_{\eta} \Lambda^{(\tau)i} = 0$. So in a step function limit, the Janus interface becomes a half-BPS superconformal interface which preserves the subalgebra $OSp(2|4)_{sc}$ of the $4d \mathcal{N} = 2$ superconformal algebra.

4.2 Off-shell construction of Janus interface in \mathbb{S}^4

In this section we explain an off-shell construction of the Janus interface on \mathbb{S}^4 . Reality properties of auxiliary fields are different from them in flat space. We also explore the relation between the Janus interface on flat space and \mathbb{S}^4 .

4.2.1 Massive subalgebra on \mathbb{S}^4

In Section 3.3.1 we explained that a half-BPS superconformal interface in a $4d \mathcal{N} = 2$ preserves the subalgebra $OSp(2|4)_{sc} \subset SU(2,2|2)$ which is a $3d \mathcal{N} = 2$ superconformal algebra. Here we explain another relevant subalgebra of SU(2,2|2), a $4d \mathcal{N} = 2$ massive subalgebra which we denote as $OSp(2|4)_{m}$. We will evaluate sphere partition function with this symmetry preserved.

4.2. CONSTRUCTION IN \mathbb{S}^4

The massive subalgebra $OSp(2|4)_{\rm m}$ is generated by Killing spinors satisfying

$$\nabla_m \epsilon^i = \frac{\mathrm{i}}{2r} \kappa^{ij} \epsilon_j, \quad \nabla_m \epsilon_i = \frac{\mathrm{i}}{2r} \overline{\kappa}_{ij} \epsilon^j, \qquad (4.17)$$

with

$$\kappa^{ij} = e^{-i\beta} (\vec{n} \cdot \vec{\tau})^{ij}, \quad \overline{\kappa}_{ij} = e^{+i\beta} (\vec{n} \cdot \vec{\tau})_{ij}.$$
(4.18)

We can diagonalize the equations (4.17) as

$$\nabla_m \chi^i = \frac{\mathrm{i}}{2r} \gamma_m \chi^i, \qquad (4.19)$$

by introducing spinors χ^i defined by

$$\chi^{i} = e^{\mathrm{i}\beta} \left(\epsilon^{i} + \kappa^{ij} \epsilon_{j} \right). \tag{4.20}$$

The above equation can be solved for ϵ^i and ϵ_i as

$$\epsilon^{i} = e^{-\frac{i}{2}\beta} P_L \chi^{i}, \quad \epsilon^{i} = e^{\frac{i}{2}\beta} (\vec{n} \cdot \vec{\tau})_{ij} P_R \chi^{j}. \tag{4.21}$$

Now we introduce the stereographic coordinates x^{μ} in which the metric is given by 3

$$g_{\mu\nu} = f(x)^2 \,\delta_{\mu\nu} \,, \qquad f(x) = \frac{1}{1 + \frac{x^2}{4r^2}} \,, \qquad (4.22)$$

where $x := \sqrt{x^{\mu}x_{\mu}}$. Then solutions of (4.19) in the stereographic coordinates are given as

$$\chi^{j} = \sqrt{f} \left(1 + \frac{\mathrm{i}}{2r} x_{\mu} \Gamma^{\mu} \right) \chi_{0}^{j} , \qquad (4.23)$$

where χ_0^j are constant spinors. By substituting this solution into (4.21), we finally obtain

$$\epsilon^{i} = e^{-\frac{\mathrm{i}}{2}\beta} \sqrt{f} \left(P_{L} \chi_{0}^{i} + \frac{\mathrm{i}}{2r} x_{\mu} \Gamma^{\mu} P_{R} \chi_{0}^{i} \right) , \qquad (4.24)$$

$$\epsilon_i = e^{\frac{i}{2}\beta} \sqrt{f} \, \vec{n} \cdot \vec{\tau}_{ij} \left(P_R \chi_0^j + \frac{i}{2r} x_\mu \Gamma^\mu P_L \chi_0^j \right) \,. \tag{4.25}$$

³Although we use the same notation, this is different from the function $f(\cdot)$ in (3.43).

We further restrict the symmetry by imposing $P_L \chi_0^i = 0$ without loss of generality ⁴. The resulting symmetry is $OSp(2|2)_m$ [29].

4.2.2 Construction

Next, we consider a construction of the Janus interface in \mathbb{S}^4 . The reality condition in this case is different from the one in flat space.

Same as in the construction in flat space, we introduce a chiral coupling multiplet $\mathcal{T} = (\tau, \Psi_i^{(\tau)}, B_{ij}^{(\tau)}, F_{\mu\nu}^{(\tau)-}, \Lambda_i^{(\tau)}, C^{(\tau)})$ and an anti-chiral coupling multiplet $\overline{\mathcal{T}} = (\overline{\tau}, \Psi^{(\tau)i}, B^{(\tau)ij}, F_{\mu\nu}^{(\tau)+}, \Lambda^{(\tau)i}, \overline{C}^{(\tau)})$. Next, we impose that $F_{ab}^{(\tau)+} = F_{ab}^{(\tau)-} = 0$ to preserve a part of Lorentz symmetry. We also impose that fermionic components in the chiral/antichiral coupling multiplet and their SUSY variations vanish to preserve a part of supersymmetry. The SUSY invariant conditions for the chiral coupling multiplet are written as

$$\delta \Psi_i^{(\tau)} = (\nabla \tau) \,\epsilon_i + \frac{1}{2} \, B_{ij}^{(\tau)} \,\epsilon^j = 0 \,, \qquad (4.27)$$

$$\delta \Lambda_i^{(\tau)} = -\frac{1}{2} \nabla B_{ij}^{(\tau)} \varepsilon^{jk} \epsilon_k + \frac{1}{2} C^{(\tau)} \varepsilon_{ij} \epsilon^j - B_{ij}^{(\tau)} \varepsilon^{jk} \eta_k = 0 .$$
(4.28)

By solving the above equations for the auxiliary fields $B_{ij}^{(\tau)}$ and $C^{(\tau)}$, we have

$$B_{ij}^{(\tau)} = \frac{4ie^{i\beta} r}{xf(x)} \tau'(x) \vec{n} \cdot \vec{\tau}_{ij} , \qquad (4.29)$$

$$C^{(\tau)} = \frac{8e^{2i\beta}r^2}{x^2 f(x)^2} \left(\tau''(x) - \frac{1}{x}\tau'(x)\right) .$$
(4.30)

Similarly, conditions for an anti-chiral coupling multiplet can be written as

$$\delta \Psi^{i} = (\nabla \overline{\tau}) \epsilon^{i} + \frac{1}{2} B^{(\tau)ij} \epsilon_{j} = 0 , \qquad (4.31)$$

$$\delta \Lambda^{i} = -\frac{1}{2} \nabla B^{(\tau) \, ij} \varepsilon_{jk} \epsilon^{k} + \frac{1}{2} \overline{C}^{(\tau)} \varepsilon^{ij} \epsilon_{j} - B^{(\tau) ij} \varepsilon_{jk} \eta^{k} = 0 \ . \tag{4.32}$$

 4 Imposing this condition corresponds to choosing a special point (north pole). If we do not impose this condition, the solution of the equation

$$\epsilon^j \propto P_L \chi^j = 0 \tag{4.26}$$

gives a special point, *i.e.* north pole. This is a system of four equations for four unknowns x^{μ} , and thus at least generically has a solution.

4.2. CONSTRUCTION IN \mathbb{S}^4

Solving the conditions (4.31) and (4.32) for $B^{(\tau)ij}$ and $\overline{C}^{(\tau)}$ leads to

$$B^{(\tau)ij} = -\frac{\mathrm{i}\,e^{-\mathrm{i}\beta}\,x}{rf(x)}\,\overline{\tau}'(x)\vec{n}\cdot\vec{\tau}^{\,ij}\,,\qquad(4.33)$$

$$\overline{C}^{(\tau)} = \frac{e^{-2i\beta} x^2}{2r^2 f(x)^2} \left(\overline{\tau}''(x) + \frac{3}{x} \overline{\tau}'(x)\right) . \tag{4.34}$$

Note that the conditions for the anti-chiral coupling multiplet (4.31)-(4.32) are formally related to those for the chiral coupling multiplet (4.27)-(4.28) by charge conjugation in Minkowski signature, but the resulting auxiliary fields do not satisfy the reality condition as (4.11) for a generic profile $\tau(x)$.

To discuss reality properties of auxiliary fields and a relation between the Janus interface in flat space and \mathbb{S}^4 , we introduce another coordinate θ which is related to stereographic coordinates by

$$x = 2r \tan \frac{\theta}{2} . \tag{4.35}$$

In this coordinate, the auxiliary fields (4.29)-(4.30) and (4.33)-(4.34) can be rewritten as

$$B_{ij}^{(\tau)} = \frac{2 \,\mathrm{i}\, e^{\mathrm{i}\beta}}{r} \cot(\theta/2) \,\frac{\mathrm{d}\tau}{\mathrm{d}\theta} \,\vec{n} \cdot \vec{\tau}_{ij} \,\,, \tag{4.36}$$

$$B^{(\tau)ij} = -\frac{2i\,e^{-i\beta}}{r}\,\tan(\theta/2)\,\frac{\mathrm{d}\overline{\tau}}{\mathrm{d}\theta}\,\vec{n}\cdot\vec{\tau}^{\,ij}\;,\qquad(4.37)$$

$$C^{(\tau)} = \frac{e^{2i\beta}}{r^2} \frac{\cos(\theta/2)}{\sin^3(\theta/2)} \left[(\cos\theta - 2) \frac{\mathrm{d}\tau}{\mathrm{d}\theta} + \sin\theta \frac{\mathrm{d}^2\tau}{\mathrm{d}\theta^2} \right] , \qquad (4.38)$$

$$\overline{C}^{(\tau)} = \frac{e^{-2i\beta}}{r^2} \frac{\sin(\theta/2)}{\cos^3(\theta/2)} \left[\left(\cos\theta + 2\right) \frac{\mathrm{d}\overline{\tau}}{\mathrm{d}\theta} + \sin\theta \frac{\mathrm{d}^2\overline{\tau}}{\mathrm{d}\theta^2} \right] .$$
(4.39)

As noted before, these auxiliary fields do not satisfy reality conditions (4.11):

$$B^{(\tau)ij} \neq (B_{ij}^{(\tau)})^*, \quad \overline{C}^{(\tau)} \neq (C^{(\tau)})^*.$$
 (4.40)

However, if we take a step function profile

$$\tau(\theta) = \begin{cases} \tau_+ & \text{for } 0 \le \theta < \frac{\pi}{2} ,\\ \tau_- & \text{for } \frac{\pi}{2} < \theta \le \pi , \end{cases}$$
(4.41)

then they satisfy the reality conditions (4.11) as in a flat space. Indeed, the auxiliary fields in θ coordinates for a step function profile are given by

$$B_{ij}^{(\tau)} = -\frac{2 \operatorname{i} e^{\mathrm{i}\beta}}{r} \vec{n} \cdot \vec{\tau}_{ij} \,\Delta\tau \,\delta\left(\theta - \frac{\pi}{2}\right) ,$$

$$C^{(\tau)} = -\frac{2 e^{2\mathrm{i}\beta}}{r^2} \,\Delta\tau \,\delta'\left(\theta - \frac{\pi}{2}\right) ,$$

$$B^{(\tau)ij} = \frac{2 \operatorname{i} e^{-\mathrm{i}\beta}}{r} \vec{n} \cdot \vec{\tau}^{ij} \,\Delta\overline{\tau} \,\delta\left(\theta - \frac{\pi}{2}\right) ,$$

$$\overline{C}^{(\tau)} = -\frac{2 e^{-2\mathrm{i}\beta}}{r^2} \,\Delta\overline{\tau} \,\delta'\left(\theta - \frac{\pi}{2}\right) ,$$
(4.42)

which satisfy the reality conditions.

Furthermore we can show that these auxiliary fields in \mathbb{S}^4 are related to them in a flat space (4.14) by the Weyl transformation and identification $e^{i\beta} = -ie^{i\alpha}$. See appendix B for more details.

4.3 Janus interface in gauge theory

In this section, we introduce the Janus interface in a $4d \mathcal{N} = 2$ gauge theory by using the construction explained in the previous section. The main tool is $\mathcal{N} = 2$ supergravity which is reviewed in Appendix A.

A general $\mathcal{N} = 2$ supersymmetric gauge theory contains a vector multiplet for a gauge group G and a matter hypermultiplet. We take an appropriate representation for a hypermultiplet to make the theory superconformal. Since the Janus interface couples to a hypermultiplet indirectly we mostly focus on a vector multiplet. We will refer to this point later.

We consider a single gauge factor with a complexified coupling constant

$$\tau = \frac{\vartheta}{2\pi} + \frac{4\pi \,\mathrm{i}}{g_{\mathrm{YM}}^2} \tag{4.43}$$

without loss of generality. The action of a vector multiplet $\mathcal{V} = (X, \Omega_i, A_\mu, Y_{ij})$ in the flat space is given as

$$I_{\text{vector}}^{\text{flat}} = \int d^4x \operatorname{Tr} \left[\frac{1}{g_{\text{YM}}^2} \left(4D_{\mu}X D^{\mu}\overline{X} - \frac{1}{2} \varepsilon^{ik} \varepsilon^{jl} Y_{ij} Y_{kl} + 2\overline{\Omega}_i D \Omega^i + \frac{1}{2} F_{\mu\nu}F^{\mu\nu} \right) + \mathrm{i} \frac{\vartheta}{16\pi^2} F_{\mu\nu} \tilde{F}^{\mu\nu} \right],$$

$$(4.44)$$

4.3. JANUS INTERFACE IN GAUGE THEORY

where $\text{Tr}[\bullet]$ is a normalized inner product of Lie algebra and D_{μ} are covariant derivatives and $\tilde{F}_{\mu\nu} = \frac{1}{2} \varepsilon_{\mu\nu}{}^{\rho\sigma} F_{\rho\sigma}$ is a dual field strength. For more details on notations and conventions, see Appendix A.3.

The action of a vector multiplet in \mathbb{S}^4 with a radius r can be obtained by performing Weyl transformation on the flat space action (4.44),

$$I_{\text{vector}} = \int d^4x \sqrt{g} \operatorname{Tr} \left[\frac{1}{g_{\text{YM}}^2} \left(4D_{\mu}X D^{\mu}\overline{X} + \frac{8}{r^2}X\overline{X} - \frac{1}{2} \varepsilon^{ik} \varepsilon^{jl} Y_{ij} Y_{kl} \right. \\ \left. + 2\overline{\Omega}_i \not\!\!\!D\Omega^i + \frac{1}{2} F_{\mu\nu}F^{\mu\nu} \right) + \mathrm{i} \frac{\vartheta}{16\pi^2} F_{\mu\nu}\tilde{F}^{\mu\nu} \right],$$

$$(4.45)$$

where $\sqrt{g} = \sqrt{-\det(g_{\mu\nu})}$. For this action to be positive semi-difinite, we demand that

$$(Y_{ij}^I)^* = -Y^{Iij} , \qquad (4.46)$$

as in [38, 39]. Note that this condition is different from the physical reality condition in Minkowski signature.

Next we explain how to write down this action via supersupace formalism. The vector multiplet \mathcal{V} can be embedded into the chiral multiplet with a Weyl weight w = 1 which we denote by $\mathcal{A}(\mathcal{V})$ as

$$A|_{\mathcal{A}(\mathcal{V})} = X,$$

$$\Psi_i|_{\mathcal{A}(\mathcal{V})} = \Omega_i ,$$

$$B_{ij}|_{\mathcal{A}(\mathcal{V})} = Y_{ij} ,$$

$$F_{ab}^{-}|_{\mathcal{A}(\mathcal{V})} = \frac{1}{2} \left(F_{ab} - \tilde{F}_{ab} \right),$$

$$\Lambda_i|_{\mathcal{A}(\mathcal{V})} = -\varepsilon_{ij} \not D\Omega^j ,$$

$$C|_{\mathcal{A}(\mathcal{V})} = \left(-2D_{\mu}D^{\mu} + \frac{4}{r^2} \right) \overline{X} .$$
(4.47)

Then we can obtain the squared chiral multiplet $\mathcal{A}(\mathcal{V})^2$ via tensor calculus which is explained in Appendix A.4. By integrating the top components of $\mathcal{A}(\mathcal{V})^2$ and its antichiral counterpart, we finally obtain the action

$$I_{\text{vector}} = \frac{1}{8\pi \,\mathrm{i}} \int \mathrm{d}^4 x \sqrt{g} \,\mathrm{Tr} \left[\tau \,C|_{\mathcal{A}(\mathcal{V})^2} - \overline{\tau} \,\overline{C}|_{\overline{\mathcal{A}}(\mathcal{V})^2} \right] \tag{4.48}$$

which is equivalent to (4.45).

Now we want to couple the gauge theory to the Janus interface defined by the coupling multiplet \mathcal{T} . The hypermultiplet does not have a explicit coupling with τ , so its interaction with the Janus interface enters indirectly via the coupling with the vector multiplet. Thus we can concentrate on the vector multiplet. To couple the Janus interface and the vector multiplet we obtain the product chiral multiplet $\mathcal{TA}(\mathcal{V})^2$ from \mathcal{T} and $\mathcal{A}(\mathcal{V})^2$ via tensor calculus. Then integrating out the top component of $\mathcal{A}(\mathcal{V})^2$ and its antichiral counterpart gives the action in the presence of the Janus interface as

$$I_{\text{Janus}} = \frac{1}{8\pi \,\mathrm{i}} \int \mathrm{d}^4 x \sqrt{g} \,\mathrm{Tr} \left[C|_{\mathcal{T}\mathcal{A}(\mathcal{V})^2} - \overline{C}|_{\overline{\mathcal{T}}\,\overline{\mathcal{A}}(\mathcal{V})^2} \right] \,. \tag{4.49}$$

Note that if we take $\tau(x)$ as a constant function, then this action reduces to the action in the absence of the Janus interface (4.45).

Chapter 5

SUSY localization and Janus interface entropy

In this chapter we calculate the sphere partition function in the presence of the Janus interface via SUSY localization. By combining this result and the formula (3.42) we show that the entropy of the Janus interface in a $4d \mathcal{N} = 2$ SCFT is proportional to Calabi's diastasis. We also have a result for the Janus interface in a $4d \mathcal{N} = 4$ super-Yang-Mills (SYM) via the AdS/CFT correspondence which gives a non-trivial check of our fourmula (3.42).

The organization of this chapter is as follows. In Section 5.1 we review the main tool for our calculation, SUSY localization. In Section 5.2 we evaluate the partition function in the presence of the Janus interface by using SUSY localization. By substituting this result into the formula (3.42), we obtain the relation between entropy of the Janus interface and Calabi's diastasis in Section 5.3. In Section 5.4 we give another calculation via the AdS/CFT correspondence.

5.1 SUSY Localization

In general to perform a path integral exactly is very hard. However, in some supersymmetric QFTs we can use a technique called the SUSY localization to perform a path integral. Let us explain this technique in this section.

Consider a supersymmetric quantum field theory with an action I_{phys} . Assume that the action is invariant under SUSY transformation: $\delta I_{\text{phys}} = 0$. The partition function is defined by

$$Z = \int [\mathrm{d}\Phi] e^{-I_{\mathrm{phys}}[\Phi]},\tag{5.1}$$

where we formally denote fields in this theory by Φ . Then we deform the original action as

$$Z(t) = \int [d\Phi] e^{-I_{\rm phys} - t\delta V}, \qquad (5.2)$$

where t is a real deformation parameter and δV is the SUSY variation of a functional V. We can show that Z(t) does not depend on t. So instead of evaluating the original path integral Z = Z(0), we can evaluate it in $t \to \infty$ limit:

$$Z(0) = Z(t) = \lim_{t \to \infty} Z(t).$$
(5.3)

By taking $t \to \infty$ we can reduce infinite dimensional integrals reduce to finite dimensional ones. This technique is called the SUSY localization.

Let us explain the result in a $4d \ \mathcal{N} = 2$ gauge theory [39]. SUSY localization reduces a path integral to the discrete sum over saddle points of δV . There are smooth saddle points and point-like (anti)instanton configurations localized at a north pole (x = 0) and a south pole $(x = \infty)$. Smooth saddle points are parametrized by a continuous variable $a \in \text{Lie } G$, while point-like (anti)instanton configurations are parametrized by two integers k, \overline{k} . The resulting integral has the form

$$Z = \int [\mathrm{d}a] e^{-I_{\mathrm{cl}}(a)} Z_{1-\mathrm{loop}}(a) Z_{\mathrm{inst}}(a,q) Z_{\mathrm{inst}}(a,\overline{q}), \qquad (5.4)$$

where

$$Z_{\text{inst}}(a,q) = \sum_{k} q^{k} Z_{k}(a), \quad Z_{\text{inst}}(a,q) = \sum_{\overline{k}} \overline{q}^{\overline{k}} Z_{\overline{k}}(a), \quad (5.5)$$

and $q = e^{2\pi i \tau}$, $\overline{q} = e^{-2\pi i \tau}$. The first factor in (5.4) $e^{-I_{cl}(a)}$ is an action evaluated at a saddle point which we denote as the classical action. The factor $Z_{1-loop}(a)$ is a contribution from fluctuations around the localization locus. Note that this factor is determined only by δV . Finally Z_{inst} are contributions from instanton/anti-instanton configurations.

46

5.2 Janus partition function

In this section we compute the partition function in the presence of the Janus interface via SUSY localization. The result for the partition function in the absence of the Janus interface is reviewed in the previous section. Now let us consider effects by introducing the Janus interface. Since the one-loop determinant depends only on δV , it is not affected by introducing the Janus interface. So we can focus on the classical action and the instanton partition function.

5.2.1 Classical action

Let us evaluate the classical action in the presence of the Janus interface. We denote the vector multiplet evaluated on the localization locus by \mathcal{V}_{cl} . The values of fields in \mathcal{V}_{cl} are given by

$$A|_{\mathcal{A}(\mathcal{V}_{\rm cl})} = X , \qquad (5.6)$$

$$B_{ij}|_{\mathcal{A}(\mathcal{V}_{cl})} = -\frac{2i e^{i\beta} X}{r} \vec{n} \cdot \vec{\tau}_{ij} , \qquad (5.7)$$

$$C|_{\mathcal{A}(\mathcal{V}_{cl})} = \frac{4 e^{2i\beta} X}{r^2} ,$$
 (5.8)

where the value of scalar component X is a constant. From this multiplet, we can compute the squared chiral multiplet $\mathcal{A}(\mathcal{V}_{cl})^2$ via tensor calculus as

$$A|_{\mathcal{A}(\mathcal{V}_{\rm cl})^2} = X^2 , \qquad (5.9)$$

$$B_{ij}|_{\mathcal{A}(\mathcal{V}_{cl})^2} = -\frac{4i e^{i\beta} X^2}{r} \vec{n} \cdot \vec{\tau}_{ij} , \qquad (5.10)$$

$$C|_{\mathcal{A}(\mathcal{V}_{\rm cl})^2} = \frac{12\,e^{2i\beta}\,X^2}{r^2}\,.\tag{5.11}$$

Then we can couple them to the Janus interface as explained in Section 4.3 to evaluate the classical action. The top component of $\mathcal{TA}(\mathcal{V}_{cl})^2$ is given by

$$C|_{\mathcal{TA}(\mathcal{V}_{cl})^2} = \frac{12 e^{2i\beta} X^2 \tau(x)}{r^2} + X^2 C^{(\tau)} + \frac{2i e^{i\beta} X^2}{r} \vec{n} \cdot \vec{\tau}^{ij} B_{ij}^{(\tau)}$$
(5.12)

$$= e^{2i\beta} X^2 \left[\frac{12}{r^2} \tau(x) + q^{(1)}(x) \tau'(x) + q^{(2)}(x) \tau''(x) \right] , \qquad (5.13)$$

where

$$q^{(1)}(x) = -\frac{8r^2}{x^3 f(x)^2} - \frac{16}{xf(x)} , \qquad q^{(2)} = \frac{8r^2}{x^2 f(x)^2} . \tag{5.14}$$

Thus the chiral part of the classical action is evaluated as

$$\int d^4x \,\sqrt{g} \,C|_{\mathcal{T}\mathcal{A}(\mathcal{V}_{\rm cl})^2} = 2\pi^2 \int_0^\infty dx \,x^3 \,f^4 \,C|_{\mathcal{T}\mathcal{A}(\mathcal{V}_{\rm cl})^2} \tag{5.15}$$

$$= 32\pi^2 e^{2i\beta} X^2 r^2 \tau(0) . \qquad (5.16)$$

The important point is that the final result only depends on the value of $\tau(x)$ at a north pole x = 0.

On the other hand, the anti-chiral multiplet from \mathcal{V}_{cl} is given as

$$\overline{A}|_{\overline{\mathcal{A}}(\mathcal{V}_{cl})} = \overline{X} , \qquad (5.17)$$

$$B^{ij}|_{\overline{\mathcal{A}}(\mathcal{V}_{cl})} = -\frac{2i\,e^{-i\beta}\,X}{r}\,\vec{n}\cdot\vec{\tau}^{\,ij}\,,\qquad(5.18)$$

$$\overline{C}|_{\overline{\mathcal{A}}(\mathcal{V}_{\rm cl})} = \frac{4 \, e^{-2i\beta} \, X}{r^2} \, . \tag{5.19}$$

By using tensor calculus we obtain

$$\int \mathrm{d}^4 x \sqrt{g} \, C|_{\overline{\mathcal{TA}}(\mathcal{V}_{\mathrm{cl}})^2} = 32\pi^2 e^{-2\mathrm{i}\beta} \, \overline{X}^2 r^2 \, \tau(\infty) \;. \tag{5.20}$$

This is again written only by the value of the coupling field $\tau(x)$ at a south pole.

The auxiliary fields in the chiral and anti-chiral multiplet from a same vector multiplet are given by a same auxiliary field \vec{Y} as

$$B_{ij} = \vec{Y} \cdot \vec{\tau}_{ij}, \quad B^{ij} = \vec{Y} \cdot \vec{\tau}^{ij}.$$

$$(5.21)$$

So by comparing (5.10) and (5.18) we can write X, \overline{X} as

$$X = \frac{1}{2}e^{-i\beta}a$$
, $\overline{X} = \frac{1}{2}e^{i\beta}a$, (5.22)

with a real constant a. The normalization for a is chosen to be consistent with [39].

5.2. JANUS PARTITION FUNCTION

Finally the classical action is given by a sum of the chiral term (5.16) and the antichiral term (5.20) with the relation (5.22):

$$I_{\rm cl}^{(\rm Janus)} = -i \pi r^2 (\tau_+ - \overline{\tau}_-) \,\mathrm{Tr} \, a^2 \,, \qquad (5.23)$$

where $\tau_+ := \tau(0)$ and $\tau_- := \tau(\infty)$. This classical action in the presence of the Janus interface can be obtained by analytically continuing moduli parameters $(\tau, \overline{\tau}) \to (\tau_+, \overline{\tau}_-)$ in the classical action in the absence of the Janus interface:

$$I_{\rm cl}^{(\rm Janus)} = I_{\rm cl}(\tau_+, \overline{\tau}_-) \ .$$
 (5.24)

5.2.2 Instanton and anti-instanton parition functions

Next we consider the instanton partition function. The instanton/antiinstanton partition functions in the absence of the Janus interface are given by (5.5). Remind that they arise from the contribution from instantons localized at a south and north pole. So an effect by introducing the Janus interface is to modify the parameters q, \bar{q} to the values of them at each poles: $q \to q_+ = e^{2\pi i \tau_+}$ and $\bar{q} \to \bar{q}_- = e^{-2\pi i \bar{\tau}_-}$. So the instanton partition function in the presence of the Janus interface are given by $Z_{\text{inst}}(a, q_+)Z_{\text{inst}}(a, \bar{q}_-)$. This result is also obtained by analytically continuing $(\tau, \bar{\tau}) \to (\tau_+, \bar{\tau}_-)$ in the result without the Janus interface similar as the classical action (5.24).

5.2.3 Partition function with the Janus interface

In summary the classical action and the instanton partition functions are obtained by an analytic continuation $(\tau, \overline{\tau}) \rightarrow (\tau_+, \overline{\tau}_-)$ in the results without the Janus interface, while the one-loop determinant is same as one in the absence of the Janus interface. Thus the resulting partition function on \mathbb{S}^4 in the presence of the Janus interface can be obtained by the same analytic continuation. On the other hand, the partition function of a $4d \mathcal{N} = 2 \text{ SCFT}$ on \mathbb{S}^4 can be written by the Kähler potential on moduli space:

$$Z_{\text{SUSY}}[\mathbb{S}^4](\tau, \overline{\tau}) = e^{K(\tau, \overline{\tau})/12} , \qquad (5.25)$$

as explained in Section 2.3.3. So by analytically continuing this expression we finally obtain the partition function with the Janus interface on \mathbb{S}^4 as

$$Z_{\text{SUSY}}^{\mathcal{I}}[\mathbb{S}^4] = e^{K(\tau_+, \bar{\tau}_-)/12} .$$
 (5.26)

5.2.4 Supergravity contenterm and Kähler ambiguity

As explained in Section 2.3.3, different UV regularization schemes cause the Kähler transformations

$$K(\tau, \overline{\tau}) \to K(\tau, \overline{\tau}) + \mathcal{F}(\tau) + \overline{\mathcal{F}}(\overline{\tau}),$$
 (5.27)

and they are explained by supergravity counterterms [26, 29] in the absence of the Janus interface. In this subsection we evaluate the supergravity counterterm which causes Kähler transformations in the presence of the Janus interface.

We can put a $4d \mathcal{N} = 2 \text{ SCFT}$ on \mathbb{S}^4 by coupling it to an off-shell Poincaré supergravity. An off-shell Poincaré supergravity can be obtained by gauge fixing an off-shell conformal supergravity using the so called compensating vector multiplet. We construct a SUSY invariant supergravity counterterm from the compensating vector multiplet according to [29]. The compensating vector multiplet on \mathbb{S}^4 which we denote by \mathcal{V}_c is given by

$$X|_{\mathcal{V}_{c}} = \mu e^{-i\beta} , \qquad Y_{ij}|_{\mathcal{V}_{c}} = -\frac{2i\mu}{r} (\vec{n} \cdot \vec{\tau})_{ij} , \qquad \Omega_{i}|_{\mathcal{V}_{c}} = F_{\mu\nu}^{-}|_{\mathcal{V}_{c}} = 0 , \quad (5.28)$$

$$\overline{X}|_{\mathcal{V}_{c}} = \mu e^{+i\beta} , \qquad Y^{ij}|_{\mathcal{V}_{c}} = -\frac{2i\mu}{r} (\vec{n} \cdot \vec{\tau})^{ij} , \qquad \overline{\Omega}^{i}|_{\mathcal{V}_{c}} = F^{+}_{\mu\nu}|_{\mathcal{V}_{c}} = 0 , \quad (5.29)$$

where $\mu > 0$ is an arbitrary mass scale. This vector multiplet can be embedded into the anti-chiral multiplet $\overline{\Phi} := \overline{\mathcal{A}}(\mathcal{V}_c)$ with Weyl weight one. We can further construct a chiral multiplet $\mathbb{T}(\log \overline{\Phi})$ with Weyl weight two from $\overline{\Phi}$. The definition of $\mathbb{T}(\log \overline{\Phi})$ is given in Appendix A.5. In this case its components are evaluated as

$$A|_{\mathbb{T}(\log\overline{\Phi})} = \frac{2\,e^{-2i\beta}}{r^2},\tag{5.30}$$

$$B_{ij}|_{\mathbb{T}(\log\overline{\Phi})} = -\frac{8\,\mathrm{i}\,e^{-\mathrm{i}\beta}}{r^3}\,(\vec{n}\cdot\vec{\tau})_{ij},\tag{5.31}$$

$$C|_{\mathbb{T}(\log \overline{\Phi})} = \frac{24}{r^4}.$$
(5.32)

Next we compute the chiral multiplet $\mathcal{F}(\mathcal{T})$ for an arbitrary holomorphic function $\mathcal{F}(\cdot)$ via the tensor calculus explained in Appendix A.4. Its

components are evaluated as

$$A|_{\mathcal{F}(\mathcal{T})} = \mathcal{F}(\tau) , \qquad (5.33)$$

$$B_{ij}|_{\mathcal{F}(\mathcal{T})} = \frac{\mathrm{d}\mathcal{F}(\tau)}{\mathrm{d}x} \frac{\mathrm{i}r \, e^{\mathrm{i}\beta}}{xf(x)} \, (\vec{n} \cdot \vec{\tau})_{ij} \,, \tag{5.34}$$

$$C|_{\mathcal{F}(\mathcal{T})} = \frac{8 r^2 e^{2i\beta}}{x^2 f^2} \left(\frac{\mathrm{d}^2 \mathcal{F}(\tau)}{\mathrm{d}x^2} - \frac{1}{x} \frac{\mathrm{d}\mathcal{F}(\tau)}{\mathrm{d}x} \right) .$$
(5.35)

According to [29] we compute the SUSY invariant counterterm from the top component of $\mathcal{F}(\mathcal{T})\mathbb{T}(\log \overline{\Phi})$. Note that the components of $\mathcal{F}(\mathcal{T})$ are obtained by replacing τ with $\mathcal{F}(\tau)$ in those of \mathcal{T} given in (4.30). Similarly the components of $\mathbb{T}(\log \overline{\Phi})$ are obtained by replacing X^2 with $\frac{2e^{-2i\beta}}{r^2}$ in those of $\mathcal{A}(\mathcal{V}_{cl})^2$ given in (5.9), (5.10), and (5.11). Therefore the top component of $\mathcal{F}(\mathcal{T})\mathbb{T}(\log \overline{\Phi})$ can be obtained from $C|_{\mathcal{T}\mathcal{A}(\mathcal{V}_{cl})^2}$ given in (5.13) by the same substitutions:

$$C\big|_{\mathcal{F}(\mathcal{T})\mathbb{T}(\log\overline{\Phi})} = \frac{2}{r^2} \left[\frac{2}{r^2} \mathcal{F}(\tau) + q^{(1)}(x) \frac{\mathrm{d}\mathcal{F}(\tau)}{\mathrm{d}x} + q^{(2)}(x) \frac{\mathrm{d}^2\mathcal{F}(\tau)}{\mathrm{d}x^2} \right] .$$
(5.36)

Thus the chiral part of SUSY invariant supergravity counterterm is given by

$$\int \mathrm{d}^4 x \,\sqrt{g} \,C|_{\mathcal{F}(\mathcal{T})\mathbb{T}(\log\overline{\Phi})} = 64\pi^2 \,\mathcal{F}(\tau_+) \,\,. \tag{5.37}$$

Similarly we can compute the anti-chiral counterterm constructed from the anti-chiral coupling multiplet $\overline{\mathcal{T}}$ and the compensating vector multiplet \mathcal{V}_c :

$$\int \mathrm{d}^4 x \sqrt{g} \,\overline{C}|_{\overline{\mathcal{F}}(\overline{\mathcal{T}})\mathbb{T}(\log \Phi)} = 64\pi^2 \,\overline{\mathcal{F}}(\overline{\tau}_-) \,\,, \tag{5.38}$$

where the anti-holomorphic $\overline{\mathcal{F}}(\overline{\tau})$ is the complex conjugate of the holomorphic function $\mathcal{F}(\tau)$ when $\overline{\tau} = \tau^*$. The supergravity conterterm is given by a sum of (5.37) and (5.38). They give rise to the Kähler transformations

$$K(\tau_+, \overline{\tau}_-) \to K(\tau_+, \overline{\tau}_-) + \mathcal{F}(\tau_+) + \overline{\mathcal{F}}(\overline{\tau}_-) ,$$
 (5.39)

in the Janus partition function $Z_{\text{SUSY}}^{\mathcal{I}}[\mathbb{S}^4]$ (5.26).

5.3 Entropy of Janus interface

By substituting the Janus partition function (5.26) into the formula (3.42), we finally obtain the entropy of the Janus interface:

$$S_{\mathcal{I}} = -\frac{1}{24} \left[K(\tau_+, \overline{\tau}_+) + K(\tau_-, \overline{\tau}_-) - K(\tau_+, \overline{\tau}_-) - K(\tau_-, \overline{\tau}_+) \right] .$$
 (5.40)

The linear combination in the bracket is Calabi's diastasis defined in Introduction. The important point is that this result does not have ambiguity due to the Kähler transformations given in (5.27) and (5.39), although the sphere partition functions given in (5.25) and (5.26) have such ambiguity.

5.4 Holographic example

In this section we focus on a theory with more supersymmetries, a $4d \mathcal{N} = 4$ super Yang-Mills (SYM) theory. A $4d \mathcal{N} = 4$ SYM theory is conjectured to be a string theory (or its supergravity limit) on a $AdS_5 \times \mathbb{S}^5$ background [40]. This correspondence is called the AdS/CFT correspondence or holography. Although there is no proof for AdS/CFT correspondence, there are many non-trivial results supporting this mysterious conjecture. This correspondence is considered to still hold in the presence of defects. In this section we compute an interface entropy and sphere partition function via holography. It provides a non-trivial check for our formula (3.42).

5.4.1 The Janus solution in supergravity

A $4d \mathcal{N} = 4$ SYM has 32 supersymmetries and an internal symmetry $SO(6)_R$. We here use the dual supergravity solution [41] to the Janus interface in the Janus interface in a $4d \mathcal{N} = 4$ SYM. The solution is given by type IIB supergravity on $AdS_4 \times \mathbb{S}^2 \times \mathbb{S}^2 \times \Sigma_2$ where Σ_2 is a two-dimensional Rimeann surface. This solution preserves the $SO(1,3) \times SO(3) \times SO(3)$ subgroup in $SO(2,3) \times SO(6)$ The metric is given by

$$ds^{2} = f_{4}^{2} ds_{AdS_{4}}^{2} + \rho^{2} dv d\overline{v} + f_{1}^{2} ds_{\mathbb{S}^{2}}^{2} + f_{2}^{2} ds_{\mathbb{S}^{2}}^{2} , \qquad (5.41)$$

where $ds_{\mathbb{S}^2}^2$ is the metric of a unit 2-sphere and v = x + iy is a complex coordinate on a strip with $x \in \mathbb{R}$ and $0 \le y \le \pi/2$. The functions f_1, f_2, f_4, ρ

5.4. HOLOGRAPHIC EXAMPLE

in the above metric are given as

$$f_4^8 = 16 \frac{F_1 F_2}{W^2} , \qquad \rho^8 = \frac{2^8 F_1 F_2 W^2}{h_1^4 h_2^4} ,$$

$$f_1^8 = 16 h_1^8 \frac{F_2 W^2}{F_1^3} , \qquad f_2^8 = 16 h_2^8 \frac{F_1 W^2}{F_2^3} ,$$
(5.42)

where

$$F_i = 2h_1 h_2 |\partial_v h_i|^2 - h_i^2 W \quad (i = 1, 2) , \qquad W = \partial_v \partial_{\overline{v}} (h_1 h_2) . \tag{5.43}$$

The two real functions $h_1(v, \overline{v})$ and $h_2(v, \overline{v})$ above are defined as

$$h_1(v,\overline{v}) = -i\alpha_1 \sinh\left(v - \frac{\Delta\phi}{2}\right) + \text{c.c.},$$
 (5.44)

$$h_2(v,\overline{v}) = \alpha_2 \cosh\left(v + \frac{\Delta\phi}{2}\right) + \text{c.c.}$$
 (5.45)

By taking $x \to \pm \infty$ we recover the dual CFT on the right/left side of the Janus interface in which we assume to have different couplings $g_{\rm YM}^{\pm}$. The dilaton ϕ in the SUGRA solution takes different values across the interface corresponding to different couplings in the dual CFT, and they are related by

$$(g_{\rm YM}^{\pm})^2 = 4\pi \left| \frac{\alpha_2}{\alpha_1} \right| e^{\pm \Delta \phi}, \qquad (5.46)$$

where $\Delta \phi$ is a difference of dilaton values and constants α_1, α_2 are related to AdS radius L as

$$L^4 = 16|\alpha_1\alpha_2|\cosh\Delta\phi. \tag{5.47}$$

5.4.2 Sphere free energy

First let us evaluate the free energy on a four-sphere. The free energy is defined as the log of a partition function. This quantity in the current setting is obtained by evaluating corresponding the on-shell supergravity action via the AdS/CFT correspondence as [42]¹

$$I = -\frac{3 \cdot 2^6 \operatorname{Vol}(\mathbb{S}^2)^2}{16\pi G_N} \int_{AdS_4} \mathrm{d}^4 x \sqrt{g_{(4)}} \int \mathrm{d}x \,\mathrm{d}y \,W h_1 h_2 , \qquad (5.48)$$

¹They evaluated this quantity in another setting.

where G_N is the Newton constant in ten dimensions and $g_{(4)}$ is a four dimensional metric of AdS_4 . To evaluate this integral, we introduce the coordinates for AdS_4 in which the metric is given by

$$\mathrm{d}s_{AdS_4}^2 = \frac{1}{\cos^2 \lambda} \left[\mathrm{d}\lambda^2 + \sin^2 \lambda \, \mathrm{d}s_{\mathbb{S}^3}^2 \right] \,, \qquad (5.49)$$

with $0 \le \lambda \le \pi/2$. Then by using these coordinates we can write the on-shell action (5.48) as

$$I = \frac{3 \operatorname{Vol}(\mathbb{S}^2)^2 \operatorname{Vol}(\mathbb{S}^3) L^8}{2^6 \pi G_N} \int_0^{\pi/2} \mathrm{d}\lambda \, \frac{\sin^3 \lambda}{\cos^4 \lambda} \\ \times \int_0^{\pi/2} \mathrm{d}y \sin^2(2y) \, \int_{-\infty}^{\infty} \mathrm{d}x \left(1 + \frac{\cosh(2x)}{\cosh(\Delta\phi)}\right) \,. \tag{5.50}$$

This integral is divergent since the AdS space has an infinite volume. So we have to regularize it. In the absence of an interface we can simply regularize the off-shell action by using Fefferman-Graham coordinates. However, in the presence of an interface, it can be difficult to construct Fefferman-Graham coordinates to cover all of the interface. One way to overcome this difficulty is to use the single cut-off regularization [43, 44]². In this approach we introduce the hypersuface defined by

$$\frac{f_4}{Z} = \frac{L}{\delta} \tag{5.51}$$

with $Z := \cos \lambda$ and eliminate the region between this hypersurface and the boundary of the AdS spacetime. In other words, the range of x is restricted to the region between $x_{\pm}(Z, y)$ which are determined by $f_4(x_{\pm}, y) = LZ/\delta$ for Z, y fixed, and the range of Z is also restricted from $Z_* := f_4(v = 0)\delta/L$ to 1. We expand $x_{\pm}(Z, y)$ in Z/δ :

$$x_{\pm}(Z, y) = \pm \frac{1}{2} \log \left(4 \cosh(\Delta \phi) \frac{Z^2}{\delta^2} \right) - \frac{\cos(2y) \tanh(\Delta \phi) \pm 2}{8} \left(\frac{\delta}{Z} \right)^2 + \mathcal{O}\left(\frac{\delta^4}{Z^4} \right).$$
(5.52)

²Another approach is to use two Fefferman-Graham coordinates and connect them to cover all space [31]. They showed that the universal part of an entanglement entropy does not depend on the detail of the curve connecting two patches. Instead one can introduce double cut-offs. This scheme was studied for the Janus interface in [44].

5.4. HOLOGRAPHIC EXAMPLE

Thus we can expand the integral over x as

$$\int_{x_{-}(Z,y)}^{x_{+}(Z,y)} \mathrm{d}x \, \left(1 + \frac{\cosh(2x)}{\cosh(\Delta\phi)}\right) \\ = \log\left(4\cosh(\Delta\phi)\frac{Z^{2}}{\delta^{2}}\right) + 2\frac{Z^{2}}{\delta^{2}} - 1 + \mathcal{O}\left(\frac{\delta^{2}}{Z^{2}}\right) \,. \tag{5.53}$$

By substituting this expansion into the on-shell action (5.50), we have

$$I = \frac{3 \operatorname{Vol}(\mathbb{S}^2)^2 \operatorname{Vol}(\mathbb{S}^3) L^8}{2^6 \pi G_N} \int_0^{\pi/2} \mathrm{d}y \, \sin^2(2y) \int_{Z_*}^1 \mathrm{d}Z \, \frac{1 - Z^2}{Z^4} \\ \times \left[\log\left(4 \cosh(\Delta \phi) \, \frac{Z^2}{\delta^2} \right) + 2 \frac{Z^2}{\delta^2} + 1 + \mathcal{O}\left(\frac{\delta^2}{Z^2}\right) \right] \quad (5.54)$$
$$= \frac{\operatorname{Vol}(\mathbb{S}^2)^2 \operatorname{Vol}(\mathbb{S}^3) L^8}{2^7 G_N} \\ \times \left[\frac{c_3}{\delta^3} + \frac{c_2}{\delta^2} + \frac{c_1}{\delta} + \log\left(\frac{4 \cosh(\Delta \phi)}{\delta^2}\right) + \frac{5}{3} + \mathcal{O}(\delta^2) \right] , \quad (5.55)$$

where c_i , (i = 1, 2, 3) are constants whose explicit forms are not needed for our discussion here.

To extract a contribution from the Janus interface, we subtract a free energy in the absence of the Janus interface which is obtained by setting $\Delta \phi = 0$ to have

$$\Delta I = I - I|_{\Delta\phi=0} = \frac{\operatorname{Vol}(\mathbb{S}^2)^2 \operatorname{Vol}(\mathbb{S}^3) L^8}{2^7 G_N} \log \cosh(\Delta\phi) .$$
 (5.56)

Note that we use the same regularization scheme for calculations of the free energy with and without the interface. Then we have the universal result, that is, the result which does not depend on the regularization scheme 3 .

The free energy evaluated in the AdS space can be mapped to one in the dual CFT. The dictionary between the deference of the value of a dilaton $\Delta \phi$ in the AdS and that of a coupling $g_{\rm YM}$ in the CFT is given in (5.46). Moreover the relation between the Newton constant and the rank of the gauge theory is given by

$$G_N = \frac{\operatorname{Vol}(\mathbb{S}^2)^2 \operatorname{Vol}(\mathbb{S}^3) L^8}{2^6 N^2}.$$
 (5.57)

³Remind that when we define an interface entropy as (3.30) we subtract the vacuum (*i.e.* no interface) contribution from the net entanglement entropy.

By using these relations, we can translate ΔI obtained from supergravity (5.56) into the CFT language as

$$\Delta I = \frac{N^2}{2} \log \left[1 + \frac{(g_{\rm YM}^+ - g_{\rm YM}^-)^2}{2g_{\rm YM}^+ g_{\rm YM}^-} \right].$$
(5.58)

So far we considered the Janus interface across which only a coupling $g_{\rm YM}$ changes its value whereas a theta angle ϑ does not. The result for the Janus interface changing a theta angle is obtained by acting $SL(2,\mathbb{R})$ transformations of type IIB supergravity on the previous result as

$$\Delta I = \frac{1}{24} \left[K(\tau_+, \overline{\tau}_+) + K(\tau_-, \overline{\tau}_-) - K(\tau_+, \overline{\tau}_-) - K(\tau_-, \overline{\tau}_+) \right] , \qquad (5.59)$$

where K is the Kähler potential given by

$$K(\tau, \overline{\tau}) = -6N^2 \log\left[i\left(\overline{\tau} - \tau\right)\right] .$$
(5.60)

5.4.3 Entanglement entropy

In this section we compute an entanglement entropy across a spherical entangling surface in the presence of the Janus interface via the AdS/CFT correspondence. This computation was done by using the Fefferman-Graham regularization in [31] and the single and double cut-off regularization in [44]. We here review the computation using single cut-off regularization [44] and compare it to the result in the previous subsection.

An entanglement entropy in a CFT in a holographic setup can be calculated by the Ryu-Takayanagi (RT) formula [45]⁴. The RT formula connects an entanglement entropy S_{Σ} across an entangling surface Σ in a CFT with the geometrical quantity in the AdS space, the area of the minimal surface anchored on the entangling surface Σ as

$$S_{\Sigma} = \min_{\gamma_{\Sigma}} \frac{\operatorname{Area}(\gamma_{\Sigma})}{4G_N},\tag{5.61}$$

where γ_{Σ} is a surface anchored on Σ . So to compute an entanglement entropy we first write down the area of a surface γ_{Σ} and then minimize it. For

⁴The formula for a CFT with a boundary was first proposed in [46].

5.4. HOLOGRAPHIC EXAMPLE

this purpose, we here move to the Lorentzian signature and introduce the Poincaré patch for AdS_4 in which the metric is given by

$$ds_{AdS_4}^2 = \frac{1}{z^2} \left[dz^2 - dt^2 + dr^2 + r^2 d\phi^2 \right] .$$
 (5.62)

In these coordinates, the spherical entangling surface we have considered is given by

$$\Sigma = \{t = 0, r = R, z = 0\} .$$
 (5.63)

Then the area of a surface γ_{Σ} in the coordinates is given by [45, 47]

$$S = \frac{\operatorname{Vol}(\mathbb{S}^2)^2 \operatorname{Vol}(\mathbb{S}^1)}{4G_N} \left(\int \mathrm{d}x \,\mathrm{d}y \,(f_1 \, f_2 \, f_4 \, \rho)^2 \right) \int \mathrm{d}z \, \frac{r}{z^2} \sqrt{1 + (\partial_z r)^2}.$$
 (5.64)

It is specified by the function r(z). Note that r(z) does not depend on (t, ϕ) because of a spherical symmetry.

Minimizing (5.64) gives the equation of motion. By solving this equation of motion we have [48]

$$r = \sqrt{R^2 - z^2} \ . \tag{5.65}$$

Similar as in the previous subsection, the integral (5.64) has divergence and we need to regularize it. To regulate this divergence due to an infinite volume of the AdS spacetime we cut it at a hypersurface specified by

$$\frac{f_4}{z} = \frac{L}{\varepsilon}.\tag{5.66}$$

Then the integration range for x is restricted to $x_{-}(z, y) \leq x \leq x_{+}(z, y)$. The values of $x_{\pm}(z, y)$ are obtained by replacing (Z, δ) with (z, ε) in (5.52). The z integral is also restricted to $z_* \equiv f_4(0)\varepsilon/L \leq z \leq R$. Then evaluating the regularized minimal area leads to

$$S = \frac{2^4 \pi \operatorname{Vol}(\mathbb{S}^2)^2 L^8}{2^4 G_N} \int_0^{\pi/2} \mathrm{d}y \, \sin^2(2y) \\ \times \int_{z_*/R}^1 \frac{\mathrm{d}z}{z^2} \int_{x_-(z,y)}^{x_+(z,y)} \mathrm{d}x \, \left(1 + \frac{\cosh(2x)}{\cosh(\Delta\phi)}\right) \,. \tag{5.67}$$

Similar as the calculation of the free energy we extract a contribution from the Janus interface by subtracting an ambient contribution:

$$S_{\mathcal{I}}|_{\text{univ}} = -\frac{N^2}{2}\log\cosh(\Delta\phi).$$
(5.68)

This again gives the universal part of the entanglement entropy. Note that this result agrees with the result obtained from another regularization [31].

Thus we show via holography that the minus of the interface entropy (5.68) is equal to the contribution to the free energy from the interface (5.56). This is consistent with our formula (3.42) derived from the CFT consideration.

Chapter 6

Conclusion and outlook

In this thesis we study the entropy of the Janus interface in a $4d \mathcal{N} = 2$ SCFT and show that it can be written by Calabi's diastasis. The former half of this thesis is on a general conformal/superconformal interface. We define an entropy of an interface as its contribution to an entanglement entropy with a spherical entangling surface. Then we derive the formula (3.42) relating the interface entropy to the sphere partition function in the presence of a half-BPS superconformal interface via the CHM map. In the intermediate step we make the assumption (3.41). We do not have a proof for it, but we give some evidences:

- It is natural in terms of the supersymmetric Rényi entropy. See Section 3.3.3.
- The results in a holographic example are consistent with the resulting formula (3.42). See Section 5.4.
- We provide a proof for the similar relation (C.8) in a $2d \mathcal{N} = (2,2)$ SCFT by using boundary super-Weyl anomaly [13]. See Appendix C.

The latter half of this thesis we focus on the Janus interface. We give an off-shell construction of the Janus interface by introducing the coupling multiplet. We then evaluate the sphere partition function in the presence of the Janus interface. By combining this result and the formula (3.42) we show that the Janus interface entropy is given by Calabi's diastasis.

There are interesting open problems. We list some of them:

- It is very important and interesting to prove (3.41). One possible way to achieve this is to use similar argument in 2d by extending the results for the boundary super-Weyl anomaly [13] to 4d.
- It is interesting to show that the contribution to the conformal anomaly from an interface indeed vanishes in the right hand side of (3.42). See Section 3.2.3.
- It can be possible to construct the Janus interface in a theory with other dimensions and SUSY. It is nice to study its entropy and investigate whether it can be written as Calabi's diastasis.
- General properties of Calabi's diastasis is interesting to study. Can we use it to investigate the structure of conformal manifold?
- General properties of an interface entropy is also interesting. Can we confirm whether it is *C*-function or not? Can we get interesting constraints on ICFTs from it?

Ultimately we want to understand general properties and structure of the space of conformal/quantum field theories. We hope that study of interface entropies can be a strong tool for this purpose.

Acknowledgement

First of all I would like to thank Takuya Okuda for collaborations and discussions. I studied a lot from his insight into physics and sincere attitude to research. I also thank Kanato Goto and Tatsuma Nishioka for collaborations and discussions related to this thesis. I thank the examiners of this thesis for useful comments on the draft. Further I thank my supervisor, Mitsuhiro Kato for his useful advice and teaching. I also thank members in Komaba particle theory group and the secretary, Toko Sasaki for their help. Finally, I would like to thank my parents for their support and encouragement.

Appendix A

Supersymmetry and supergravity

A.1 Notations and conventions

Here We mostly use conventions and notations used in [49, 25]. The imaginary unit is denoted by i. Complex conjugation is indecated by * while hermitian conjugation by \dagger .

A.1.1 Signature and coordinate index

We denote coordinates indices on a general manifold as μ, ν, \cdots while coordinates indices on a flat space as a, b, \cdots . Indices run within 0, 1, 2, 3 in the Lorentzian signature, while 1, 2, 3, 4 in the Euclidean signature. A flat space metric is given by $\eta^{ab} = \text{diag}(-1, +1, +1, +1)$ in the Lorentzian signature and $\eta^{ab} = \text{diag}(+1, +1, +1, +1)$ in the Euclidean signature. The vielbein and its inverse are denoted by $e_{\mu}{}^{a}$ and $e_{a}{}^{\mu}$ respectively.

A.1.2 Gamma matrix

The gamma matrices γ^{μ} with a Greek alphabet satisfy

$$\{\gamma^{\mu}, \gamma^{\nu}\} = 2g^{\mu\nu} , \qquad (A.1)$$

while the gamma matrices γ^{μ} with a Latin alphabet satisfy

$$\{\gamma^a, \gamma^b\} = 2\eta^{ab} . \tag{A.2}$$

We sometimes denote γ^a as Γ^a . We can take a specific matrix representation to have more explicit expressions. For example, in the Weyl representation, γ^a are given by

$$\gamma^a = \begin{pmatrix} 0 & \sigma^\mu \\ \overline{\sigma}^\mu & 0 \end{pmatrix}, \tag{A.3}$$

where $\sigma^{\mu} = (\sigma^1, \sigma^2, \sigma^3, i), \overline{\sigma}^a = (\sigma^1, \sigma^2, \sigma^3, -i)$, and $\sigma^i (i = 1, 2, 3)$ are Pauli matrices.

The gamma matrices with Greek and Latin indices are related by vielbein as

$$\gamma^{\mu} = \gamma^a e_a{}^{\mu} . \tag{A.4}$$

The gamma matrix γ^a is hermitian if a = 0 and anti-hermitian otherwise. The gamma matrices in the Lorentzian signature and in the Euclidean signature are related by

$$\gamma^{a=0} = -i\gamma^{a=4},\tag{A.5}$$

and the other components are same.

We define the chirality matrix by $\gamma_* = i\gamma_0\gamma_1\gamma_2\gamma_3 = \gamma_1\gamma_2\gamma_3\gamma_4$. By using the matrix γ_* , we define chiral projection operators as

$$P_L = \frac{1}{2} (1 + \gamma_*) , \qquad P_R = \frac{1}{2} (1 - \gamma_*) .$$
 (A.6)

A.1.3 Charge and Weyl conjugation

The charge conjugate matrix C satisfies

$$CC^{\dagger} = 1 \quad C^{T} = -C , \quad C\gamma_{\mu}C^{-1} = -\gamma_{\mu}^{T} .$$
 (A.7)

Then the charge conjugation of a four-component spinor Ψ which we denote as Ψ^C is defined by

$$\Psi^C = B^{-1} \Psi^* , \qquad (A.8)$$

where $B = i C \gamma^0$. By definition the matrix B satisfies the relation

$$B^{-1}(\gamma^{\mu})^* B = \gamma^{\mu} .$$
 (A.9)

We list some properties of the charge conjugation:

$$(\Psi^C)^C = \Psi, \quad (\gamma_{\mu_1} \dots \gamma_{\mu_N} \Psi)^C = \gamma_{\mu_1} \dots \gamma_{\mu_N} \Psi^C,$$
 (A.10)

A.2. SUPERSYMMETRY PARAMETERS

and

$$(\overline{\epsilon}\gamma_{\mu_1}\dots\gamma_{\mu_N}\eta)^* = \pm \overline{\epsilon^C}\gamma_{\mu_1}\dots\gamma_{\mu_N}\eta^C .$$
 (A.11)

We denote the Weyl conjugate of Ψ as $\overline{\Psi}$ which is defined by

$$\overline{\Psi} := \Psi^T C . \tag{A.12}$$

Note that this is equivalent to the Dirac conjugate when the spinor satisfies the certain condition. See Appendix A.2.

A.1.4 $SU(2)_R$ multiplets

We denote $SU(2)_R$ doublet indices by i, j, \ldots . We can regard $SU(2)_R$ triplet as a tree-component vector \vec{Y} . We can construct the 2×2 matrix Y_i^j from \vec{Y} as

$$Y_i^j = \vec{\tau}_i^{\ j} \cdot \vec{Y} , \qquad (A.13)$$

where $\vec{\tau}_i^{\ j} = i \vec{\sigma}_i^{\ j}$. $SU(2)_R$ indices are raised or lowered by anti-symmetric tensors ε^{ij} and ε_{ij} which satisfy $\varepsilon^{12} = \varepsilon_{12} = 1$ as

$$\vec{\tau}^{ij} = \varepsilon^{ik} \vec{\tau}_k^{\ j} = (\vec{\tau}_{ij})^* = \varepsilon^{ik} \varepsilon^{jl} \vec{\tau}_{kl} \ . \tag{A.14}$$

By using τ^{ij} we can also construct a symmetric matrix Y^{ij} from \vec{Y} as

$$Y^{ij} = \vec{\tau}^{\,ij} \cdot \vec{Y} \ . \tag{A.15}$$

A.2 Supersymmetry parameters

The Poincaré SUSY parameters ϵ^i, ϵ_i in the Lorentzian signature satisfy

$$(\epsilon^i)^C = \epsilon_i . \tag{A.16}$$

For spinors satisfying the above condition, the Weyl conjugate is equivalent to the Dirac conjugate:

$$\bar{\epsilon}^i = (\epsilon_i)^\dagger i \gamma^0 . \tag{A.17}$$

The special conformal SUSY parameters η^i, η_i satisfy the same condition as

$$(\eta^i)^C = \eta_i . \tag{A.18}$$

According to [25], these parameters are chiral in the Lorentzian and Euclidean signature :

$$\epsilon^i = P_L \epsilon^i$$
, $\epsilon_i = P_R \epsilon_i$, $\eta^i = P_R \eta^i$, $\eta_i = P_L \eta_i$. (A.19)

A.3 $\mathcal{N} = 2$ supermultiplets

In the rest of Appendix A, we assume that the values of the fields in the Weyl multiplet vanish except for a metric and vielbein. We also assume that the SUSY parameters satisfy the following conditions [29]:

$$\eta^{i} = \frac{1}{4} \gamma^{\mu} \nabla_{\mu} \epsilon^{i}, \quad \eta_{i} = \frac{1}{4} \gamma^{\mu} \nabla_{\mu} \epsilon_{i}.$$
 (A.20)

Under these assumptions, we give the SUSY transformations of a vector and chiral multiplet.

A.3.1 Vector multiplet

A vector multiplet has $(X, \Omega_i, A_\mu, Y_{ij})$ as its components. The spinor Ω_i is left-chiral and its conjugate Ω_i is right-chiral. They are related by $(\Omega_i^I)^C = \Omega^{Ii}$ in Minkowski space. We expand fields in terms of hermitian generators T_I as $X = T_I X^I$, $A_\mu = T_I A_\mu^I$, and so on. Then we give the SUSY transformations for them as [49]

$$\delta A^{I}_{\mu} = \frac{1}{2} \,\varepsilon^{ij} \,\overline{\epsilon}_{i} \gamma_{\mu} \Omega^{I}_{j} + \frac{1}{2} \,\varepsilon_{ij} \,\overline{\epsilon}^{i} \gamma_{\mu} \Omega^{jI} \,\,, \tag{A.23}$$

A.3.2 Chiral multiplet

A chiral multiplet \mathcal{A} has $(A, \Psi_i, B_{ij}, F_{ab}^-, \Lambda_i, C)$ as its components. Their SUSY transformations are [50, 49]

$$\delta A = \frac{1}{2} \,\overline{\epsilon}^i \Psi_i \,\,, \tag{A.25}$$

$$\delta \Psi_i = \nabla (A \epsilon_i) + \frac{1}{2} B_{ij} \epsilon^j + \frac{1}{4} \Gamma^{ab} F^-_{ab} \varepsilon_{ij} \epsilon^j + (2w - 4) A \eta_i , \qquad (A.26)$$

$$\delta B_{ij} = \overline{\epsilon}_{(i} \nabla \Psi_{j)} - \overline{\epsilon}^k \Lambda_{(i} \varepsilon_{j)k} + 2(1-w) \overline{\eta}_{(i} \Psi_{j)} , \qquad (A.27)$$

$$\delta F_{ab}^{-} = \frac{1}{4} \varepsilon^{ij} \overline{\epsilon}_i \, \nabla \Gamma_{ab} \, \Psi_j + \frac{1}{4} \overline{\epsilon}^i \, \Gamma_{ab} \Lambda_i - \frac{1}{2} \left(1 + w \right) \varepsilon^{ij} \, \overline{\eta}_i \, \Gamma_{ab} \Psi_j \,, \qquad (A.28)$$

$$\delta \Lambda_i = -\frac{1}{4} \Gamma^{ab} \nabla (F_{ab}^- \epsilon_i) - \frac{1}{2} \nabla B_{ij} \varepsilon^{jk} \epsilon_k + \frac{1}{2} C \varepsilon_{ij} \epsilon^j - (1+w) B_{ij} \varepsilon^{jk} \eta_k + \frac{1}{2} (3-w) \Gamma^{ab} F_{ab}^- \eta_i , \qquad (A.29)$$

$$\delta C = -\nabla_{\mu} (\varepsilon^{ij} \,\overline{\epsilon}_i \,\gamma^m \Lambda_j) + (2w - 4) \,\varepsilon^{ij} \,\overline{\eta}_i \,\Lambda_j \,\,, \tag{A.30}$$

where w is the Weyl weight of the multiplet.

An anti-chiral multiplet $\overline{\mathcal{A}}$ has $(\overline{A}, \Psi^{i}, B^{ij}, F^{+}_{ab}, \Lambda^{i}, \overline{C})$ as its components. The SUSY transformations in Lorentzian signature are obtained by taking complex/Weyl conjugate of SUSY transformations of a chiral multiplet. The transformations in Euclidean signature are formally given by the same procedure.

A.4 Tensor calculus for chiral multiplets

Let us consider two chiral multiplets \mathcal{A}, \mathcal{B} whose fermionic components vanish

$$\mathcal{A} = (A|_{\mathcal{A}}, \Psi_i|_{\mathcal{A}} = 0, B_{ij}|_{\mathcal{A}}, F_{ab}^-|_{\mathcal{A}}, \Lambda_i|_{\mathcal{A}} = 0, C|_{\mathcal{A}}), \qquad (A.31)$$

$$\mathcal{B} = (A|_{\mathcal{B}}, \Psi_i|_{\mathcal{B}} = 0, B_{ij}|_{\mathcal{B}}, F_{ab}^-|_{\mathcal{B}}, \Lambda_i|_{\mathcal{B}} = 0, C|_{\mathcal{B}}) .$$
(A.32)

Then we can make the product of these chiral multiplet which is also a chiral multiplet as [51]

$$A|_{\mathcal{AB}} = A|_{\mathcal{A}} A|_{\mathcal{B}} , \qquad (A.33)$$

$$B_{ij}|_{\mathcal{AB}} = A|_{\mathcal{A}} B_{ij}|_{\mathcal{B}} + A|_{\mathcal{B}} B_{ij}|_{\mathcal{A}} , \qquad (A.34)$$

$$F_{ab}^{-}|_{\mathcal{AB}} = A|_{\mathcal{A}} F_{ab}^{-}|_{\mathcal{B}} + A|_{\mathcal{B}} F_{ab}^{-}|_{\mathcal{A}} , \qquad (A.35)$$

$$C|_{\mathcal{A}\mathcal{B}} = A|_{\mathcal{A}} C|_{\mathcal{B}} + C|_{\mathcal{A}} A|_{\mathcal{B}} - \frac{1}{2} \varepsilon^{ik} \varepsilon^{jl} B_{ij}|_{\mathcal{A}} B_{kl}|_{\mathcal{B}} + F_{ab}^{-}|_{\mathcal{A}} F^{-ab}|_{\mathcal{B}} .$$
(A.36)

By iterating the above product, we obtain the n-th product of a chiral multiplet as

$$A|_{\mathcal{A}^n} = \left(A|_{\mathcal{A}}\right)^n \,, \tag{A.37}$$

$$B_{ij}|_{\mathcal{A}^n} = n \left(A|_{\mathcal{A}}\right)^{n-1} B_{ij}|_{\mathcal{A}} , \qquad (A.38)$$

$$F_{ab}^{-}|_{\mathcal{A}^{n}} = n \left(A|_{\mathcal{A}}\right)^{n-1} F_{ab}^{-}|_{\mathcal{A}} , \qquad (A.39)$$

$$C|_{\mathcal{A}^{n}} = n \left(A|_{\mathcal{A}}\right)^{n-1} C|_{\mathcal{A}} - \frac{1}{4} n(n-1) \left(A|_{\mathcal{A}}\right)^{n-2} \left[\varepsilon^{ik} \varepsilon^{jl} B_{ij}|_{\mathcal{A}} B_{kl}|_{\mathcal{A}} - 2 \left(F_{ab}^{-}|_{\mathcal{A}}\right)^{2}\right]$$
(A.40)

We have to apply the above formulae to the coefficients of generators T_I when fields are in the adjoint representation.

A.5 Definition of $\mathbb{T}(\log \overline{\Phi})$

In this appendix we give the definition of the chiral multiplet $\mathbb{T}(\log \overline{\Phi})$ made from an anti-chiral multiplet $\overline{\Phi}$ with vanishing fermionic and field strength components. First, the components of $\log \overline{\Phi}$ are given by [52]

$$\overline{A}|_{\log \overline{\Phi}} = \log\left(\overline{A}|_{\overline{\Phi}}\right),\tag{A.41}$$

$$B^{ij}|_{\log \overline{\Phi}} = \frac{B^{ij}|_{\overline{\Phi}}}{\overline{A}|_{\overline{\Phi}}},\tag{A.42}$$

$$\overline{C}|_{\log \overline{\Phi}} = \frac{\overline{C}|_{\overline{\Phi}}}{\overline{A}|_{\overline{\Phi}}} + \frac{1}{4\left(\overline{A}|_{\overline{\Phi}}\right)^2} \varepsilon_{ik} \varepsilon_{jl} \left(B^{ij}|_{\overline{\Phi}}\right) \left(B^{kl}|_{\overline{\Phi}}\right).$$
(A.43)

The chiral multiplet $\mathbb{T}(\text{anti-chiral multiplet})$ is the so-called $\mathcal{N} = 2$ kinetic multiplet [53]. The components of the kinetic multiplet made from $\log \overline{\Phi}$ are then given as [52]

$$A|_{\mathbb{T}(\log\overline{\Phi})} = \overline{C}|_{\log\overline{\Phi}} , \qquad (A.44)$$

$$B_{ij}|_{\mathbb{T}(\log\overline{\Phi})} = -2\varepsilon_{ik}\varepsilon_{jl}\Box_C B^{kl}|_{\log\overline{\Phi}} , \qquad (A.45)$$

$$C|_{\mathbb{T}(\log \overline{\Phi})} = 4 \Box_C \Box_C \overline{A}|_{\log \overline{\Phi}} , \qquad (A.46)$$

where \Box_C is the so-called conformal d'Alembertian.

A.6 Supercurrent multiplet

The components of the supercurrent multiplet for an Abelian vector multiplet can be obtained by linearizing the superconformal action (20.89) of [25] in terms of the Weyl multiplet. They are given by

$$T^{\mu\nu} = 8\partial^{(\mu}\overline{X}\partial^{\nu)}X - 4g^{\mu\nu}\left|\partial_{\rho}X\right|^{2} + \frac{4}{3}(g^{\mu\nu}\partial^{2} - \partial^{\mu}\partial^{\nu})\left|X\right|^{2} - g^{\mu\nu}(\overline{X}\partial^{2}X + X\partial^{2}\overline{X}) + \overline{\Omega}^{i}\gamma^{(\mu}\overleftrightarrow{\partial}^{\nu)}\Omega_{i} - \frac{1}{4}g^{\mu\nu}\overline{\Omega}^{i}\overleftrightarrow{\mathcal{D}}\Omega_{i} + 2F^{\mu}{}_{\rho}F^{\nu\rho} - \frac{1}{2}g^{\mu\nu}F_{\rho\sigma}F^{\rho\sigma},$$
(A.47)

$$S_{i}^{\mu} = -\frac{1}{2} F_{\rho\sigma} \gamma^{\rho\sigma} \gamma^{\mu} \varepsilon_{ij} \Omega^{j} - 2\overline{X} \overset{\leftrightarrow}{\partial}^{\mu} \Omega_{i} + 2\overline{X} \gamma^{\mu} \partial \Omega_{i} - \frac{2}{3} \gamma^{\mu\nu} \partial_{\nu} \left(\overline{X} \Omega_{i} \right), \quad (A.48)$$

$$j_{\mu}{}^{i}{}_{j} = -2\overline{\Omega}{}^{i}\gamma_{\mu}\Omega_{j} + \delta^{i}{}_{j}\overline{\Omega}{}^{k}\gamma_{\mu}\Omega_{k}, \quad j_{\mu} = -4\mathrm{i}\overline{X}\overline{\overleftrightarrow{\partial}}_{\mu}X + \mathrm{i}\overline{\Omega}{}^{i}\gamma_{\mu}\Omega_{i}, \tag{A.49}$$

$$J = -4XX, \quad j_i = 4X\Omega_i, \quad j_{\mu\nu}^+ = XF_{\mu\nu}^+, \quad j_{\mu\nu}^- = XF_{\mu\nu}^-.$$
(A.50)

They are equivalent to the supercurrent multiplet written in [54] in terms of two component notations.

Appendix B

Weyl transformation between \mathbb{S}^4 and flat space

In this appendix we show that the auxiliary fields in the coupling multiplet in a flat space and \mathbb{S}^4 are related by a Weyl transformation.

We denote Cartesian coordinates in a flat space by y^{μ} while stereographic coordinates for \mathbb{S}^4 by x^{μ} . For this purpose, we introduce the five-dimensional embedding coordinates Y^M ($M = 1, \dots, 5$). They satisfy

$$\sum (Y^M)^2 = r^2$$
, $ds_{\mathbb{S}^4}^2 = \sum (dY^M)^2$. (B.1)

They can be parametrized by x and y-coordinates as

$$\begin{pmatrix} Y^{1} \\ Y^{2} \\ Y^{3} \\ Y^{4} \\ Y^{5} \end{pmatrix} = \begin{pmatrix} f(x) \begin{pmatrix} x^{1} \\ x^{2} \\ x^{3} \\ x^{4} \end{pmatrix} \\ g(x) \end{pmatrix} = \begin{pmatrix} g(y) \\ \begin{pmatrix} y^{4} \\ y^{1} \\ y^{2} \\ y^{3} \end{pmatrix} \end{pmatrix}, \quad (B.2)$$

where the function $f(\cdot)$ is defined in (4.22) and $g(\cdot)$ is defined by

$$g(z) := r \frac{1 - \frac{z^2}{4r^2}}{1 + \frac{z^2}{4r^2}} .$$
(B.3)

By using these relations, we can relate the delta function in θ and y-coordinates as

$$\frac{1}{r}\delta\left(\theta - \frac{\pi}{2}\right) = \frac{1}{f(y)}\,\delta(y^3)\,,\qquad \frac{1}{r^2}\delta'\left(\theta - \frac{\pi}{2}\right) = -\frac{1}{f(y)^2}\,\delta'(y^3)\,.\tag{B.4}$$

By substituting these relations into (4.14) and (4.42) we obtain the relations between them as

$$B_{ij}^{(\tau)}|_{\mathbb{S}^4} = \frac{1}{f(y)} B_{ij}^{(\tau)}|_{\mathbb{R}^4}, \quad B^{(\tau)ij}|_{\mathbb{S}^4} = \frac{1}{f(y)} B^{(\tau)ij}|_{\mathbb{R}^4}, \tag{B.5}$$

$$C^{(\tau)}|_{\mathbb{S}^4} = \frac{1}{(f(y))^2} C^{(\tau)}|_{\mathbb{R}^4}, \quad \overline{C}|_{\mathbb{S}^4} = \frac{1}{(f(y))^2} \overline{C}|_{\mathbb{R}^4}.$$
 (B.6)

Note that the function f(y) is the Weyl factor of transformations between the flat space and \mathbb{S}^4 and the powers of f(y) in the above relations correspond to the Weyl weights of the auxiliary fields $B_{ij}^{(\tau)}, B^{(\tau)ij}, C^{(\tau)}, \overline{C}$ which are 1, 1, 2, 2, respectively. Thus we have shown that the auxiliary fields in a flat space and \mathbb{S}^4 are related by the Weyl transformation.
Appendix C

Boundary super-Weyl anomaly in $2d \mathcal{N} = (2, 2)$ SCFT

In this appendix we give a proof for the similar statement as (3.41) in 2d $\mathcal{N} = (2,2)$ SCFTs by using the boundary super-Weyl anomaly [13].

C.1 Super-Weyl anomaly on closed manifold

First we review the result in [55]. The authors of the paper considered the supersymmetric version of the Weyl anomaly called the super-Weyl anomaly in $2d \mathcal{N} = (2, 2)$ SCFTs on a closed Riemann manifold.

The Weyl anomaly is defined by rescaling a metric $g_{\mu\nu}$. Similarly, the super-Weyl anomaly is defined by varying values of the components in a graviton multiplet. We take a conformal gauge $g_{\mu\nu} = e^{2\sigma} \delta_{\mu\nu}$ and introduce Cartesian coordinates (x^1, x^2) . We then consider the so-called $U(1)_V$ supergravity in which the graviton multiplet is a twisted chiral multiplet Σ . The lowest component of Σ is the combination of a metric $g_{\mu\nu}$ and a gauge field V^{μ} that couples to the vector-like *R*-symmetry:

$$\Sigma(y^{\mu}) = \sigma(y^{\mu}) + \mathrm{i}\,a(y^{\mu}) + \theta^{+}\overline{\chi}_{+}(y^{\mu}) + \overline{\theta}^{-}\chi_{-}(y^{\mu}) + \theta^{+}\overline{\theta}^{-}w(y^{\mu}), \qquad (\mathrm{C.1})$$

where $y^{\pm} = x^{\pm} \mp \theta^{\pm} \overline{\theta}^{\pm}$ and $x^{\pm} = x^1 \pm x^2$. See Chapter 12 of [56] for the superfield conventions in $2d \mathcal{N} = (2,2)$ SCFTs. Then the anomalous dependence of the partition function on the graviton multiplet Σ for a closed manifold \mathcal{M} is given by [55]

$$A_{\text{closed}} := -\mathrm{i}\delta_{\Sigma}\log Z[\mathcal{M}], \qquad (C.2)$$
$$= -\frac{\mathrm{i}}{4\pi}\int_{\mathcal{M}}\mathrm{d}^{2}x\int\mathrm{d}^{4}\theta\left[\frac{c}{6}(\delta\Sigma\overline{\Sigma}+\delta\overline{\Sigma}\Sigma)-(\delta\Sigma+\delta\overline{\Sigma})K(\mathcal{T},\overline{\mathcal{T}})\right], (C.3)$$

where \mathcal{T} is a twisted chiral multiplet whose lowest component is a moduli parameter τ and the function $K(\tau, \overline{\tau})$ is identified with the Kähler potential on the moduli space. By integrating (C.2), we have

$$\log Z[\mathcal{M}] \supset \frac{1}{4\pi} \int_{\mathcal{M}} \mathrm{d}^2 x \int \mathrm{d}^4 \theta \left[\frac{c}{6} \Sigma \overline{\Sigma} - (\Sigma + \overline{\Sigma}) K(\mathcal{T}, \overline{\mathcal{T}}) \right].$$
(C.4)

The first term gives the contribution from the usual Weyl anomaly, while the second term gives the contribution as (2.34).

C.2 Super-Weyl anomaly on open manifold and hemisphere partition function

The author of the paper [13] extend the result reviewed in the previous section to the super-Weyl anomaly for $2d \mathcal{N} = (2,2)$ SCFTs with a supersymmetric boundary. We here review their results and then prove the similar statement as (3.41) for $2d \mathcal{N} = (2,2)$ SCFTs by using their result.

The author of the paper [13] showed that the super-Weyl anomaly on a manifold \mathcal{M} with a supersymmetric boundary is given by

$$A_{\text{open}} := -\mathrm{i}\delta_{\Sigma}\log Z[\mathcal{M}], \qquad (C.5)$$
$$\supset \delta \Big[\frac{\mathrm{i}}{4\pi} \int_{\mathcal{M}} \mathrm{d}^{2}x \left(\Box(\sigma - \mathrm{i}\,a)h^{\Omega} + \Box(\sigma + \mathrm{i}\,a)\overline{h}^{\Omega}\right) + \frac{1}{4\pi} \int_{\partial\mathcal{M}} \mathrm{d}x^{2}(\overline{w}h^{\Omega} - w\overline{h}^{\Omega})\Big], \qquad (C.6)$$

where dropped moduli independent terms and terms that contain derivatives of moduli parameters. The function $h^{\Omega}(\tau)$ is related to the holomorphic central charge $c^{\Omega}(\tau)$ which gives the hemisphere partition function [57, 58, 59] as

$$c^{\Omega} = \exp h^{\Omega}. \tag{C.7}$$

We call the anomaly (C.5) as the boundary super-Weyl anomaly.

C.2. BOUNDARY SUPER-WEYL ANOMALY

Now, we consider the boundary super-Weyl anomaly on a two-dimensional hemisphere D^2 and prove that the hemisphere partition function computed in a conformal scheme $Z^{(\text{BCFT})}[D^2]$ and in a supersymmetric scheme $Z^{\mathcal{B}}_{\text{SUSY}}[D^2]$ are related by

$$Z^{(\text{BCFT})}[D^2] = \left| Z^{\mathcal{B}}_{\text{SUSY}}[D^2] \right|.$$
(C.8)

This is a 2d boundary CFT version of the statement (3.41).

The D^2 background is given by

$$\sigma = -\log(1+z\overline{z}), \quad a = 0, \tag{C.9}$$

where $z = x^1 + ix^2$. First we want to evaluate the partition function in a conformal scheme which instead violates supersymmetry. For this purpose, we turn off auxiliary fields in a Weyl multiplet: $w = \overline{w} = 0$. Then integrating (C.5) leads to

$$Z^{(\text{BCFT})}[D^2] = Z_0 \exp\left[\frac{1}{2}(h^{\Omega}(\tau) + \overline{h}^{\Omega}(\overline{\tau}))\right], \qquad (C.10)$$

where Z_0 is a moduli independent factor due to the Weyl anomaly ¹.

On the other hand, if we want to evaluate the partition function in a SUSY preserving scheme which instead violates the conformal symmetry, we take the background as

$$w = \overline{w} = -\frac{2\mathbf{i}}{1+z\overline{z}}.\tag{C.11}$$

Then the resulting partition function is

$$Z_{\text{SUSY}}^{\mathcal{I}}[D^2] = Z_0 \exp\left[h^{\Omega}(\tau)\right] = Z_0 c^{\Omega}(\tau).$$
 (C.12)

This is the hemisphere partition function obtained by [57, 58, 59]. Thus we prove the similar statement as (3.41) in a 2d $\mathcal{N} = (2, 2)$ SCFT with a supersymmetric boundary.

¹See (2.34) and the footnote 9 in Section 2.3.3.

Bibliography

- [1] P. Liendo, L. Rastelli and B. C. van Rees, *The Bootstrap Program for* Boundary CFT_d , JHEP **07** (2013) 113, [1210.4258].
- [2] N. Kobayashi, T. Nishioka, Y. Sato and K. Watanabe, Towards a C-theorem in defect CFT, JHEP 01 (2019) 039, [1810.06995].
- [3] A. Zamolodchikov, Irreversibility of the Flux of the Renormalization Group in a 2D Field Theory, JETP Lett. 43 (1986) 730–732.
- [4] Z. Komargodski and A. Schwimmer, On Renormalization Group Flows in Four Dimensions, JHEP 12 (2011) 099, [1107.3987].
- [5] R. C. Myers and A. Sinha, Holographic c-theorems in arbitrary dimensions, JHEP 01 (2011) 125, [1011.5819].
- [6] D. L. Jafferis, I. R. Klebanov, S. S. Pufu and B. R. Safdi, Towards the F-Theorem: N = 2 Field Theories on the Three-Sphere, JHEP 06 (2011) 102, [1103.1181].
- [7] H. Casini and M. Huerta, On the RG running of the entanglement entropy of a circle, Phys. Rev. D 85 (2012) 125016, [1202.5650].
- [8] D. Bak, M. Gutperle and S. Hirano, A Dilatonic deformation of AdS(5) and its field theory dual, JHEP 05 (2003) 072, [hep-th/0304129].
- [9] A. B. Clark, D. Z. Freedman, A. Karch and M. Schnabl, Dual of the Janus solution: An interface conformal field theory, Phys. Rev. D71 (2005) 066003, [hep-th/0407073].
- [10] C. P. Bachas, I. Brunner, M. R. Douglas and L. Rastelli, *Calabi's diastasis as interface entropy*, *Phys. Rev.* D90 (2014) 045004, [1311.2202].

- [11] E. Calabi, Isometric imbedding of complex manifolds, Annals of Mathematics 58 (1953) 1–23.
- [12] E. D'Hoker and M. Gutperle, Holographic entropy and Calabi's diastasis, JHEP 10 (2014) 093, [1406.5124].
- [13] C. Bachas and D. Plencner, Boundary Weyl anomaly of $\mathcal{N} = (2, 2)$ superconformal models, JHEP **03** (2017) 034, [1612.06386].
- [14] K. Goto and T. Okuda, Janus interface in two-dimensional supersymmetric gauge theories, JHEP 10 (2019) 045, [1810.03247].
- [15] K. Goto, L. Nagano, T. Nishioka and T. Okuda, Janus interface entropy and Calabi's diastasis in four-dimensional $\mathcal{N} = 2$ superconformal field theories, JHEP **08** (2020) 048, [2005.10833].
- [16] K. Goto and T. Okuda, Interface entropy in four dimensions as Calabi['] s diastasis on the conformal manifold, JHEP **11** (2018) 122, [1805.09981].
- [17] D. Gaiotto and E. Witten, Janus Configurations, Chern-Simons Couplings, And The theta-Angle in N = 4 Super Yang-Mills Theory, JHEP 06 (2010) 097, [0804.2907].
- [18] E. D'Hoker, J. Estes and M. Gutperle, Interface Yang-Mills, supersymmetry, and Janus, Nucl. Phys. B 753 (2006) 16–41, [hep-th/0603013].
- [19] C. Kim, E. Koh and K.-M. Lee, Janus and Multifaced Supersymmetric Theories, JHEP 06 (2008) 040, [0802.2143].
- [20] C. Kim, E. Koh and K.-M. Lee, Janus and Multifaced Supersymmetric Theories II, Phys. Rev. D 79 (2009) 126013, [0901.0506].
- [21] N. Drukker, D. Gaiotto and J. Gomis, The Virtue of Defects in 4D Gauge Theories and 2D CFTs, JHEP 06 (2011) 025, [1003.1112].
- [22] H. Casini, M. Huerta and R. C. Myers, Towards a derivation of holographic entanglement entropy, JHEP 05 (2011) 036, [1102.0440].

- [23] P. Di Francesco, P. Mathieu and D. Senechal, *Conformal Field Theory*. Graduate Texts in Contemporary Physics. Springer-Verlag, New York, 1997, 10.1007/978-1-4612-2256-9.
- [24] D. Simmons-Duffin, The Conformal Bootstrap, in Theoretical Advanced Study Institute in Elementary Particle Physics: New Frontiers in Fields and Strings, pp. 1–74, 2017. 1602.07982. DOI.
- [25] D. Z. Freedman and A. Van Proeyen, *Supergravity*. Cambridge Univ. Press, Cambridge, UK, 2012.
- [26] E. Gerchkovitz, J. Gomis and Z. Komargodski, Sphere Partition Functions and the Zamolodchikov Metric, JHEP 11 (2014) 001, [1405.7271].
- [27] H. Jockers, V. Kumar, J. M. Lapan, D. R. Morrison and M. Romo, Two-Sphere Partition Functions and Gromov-Witten Invariants, Commun. Math. Phys. 325 (2014) 1139–1170, [1208.6244].
- [28] J. Gomis and S. Lee, Exact Kähler Potential from Gauge Theory and Mirror Symmetry, JHEP 04 (2013) 019, [1210.6022].
- [29] J. Gomis and N. Ishtiaque, Kähler potential and ambiguities in 4d \mathcal{N} = 2 SCFTs, JHEP **04** (2015) 169, [1409.5325].
- [30] M. Billò, V. Gonçalves, E. Lauria and M. Meineri, Defects in conformal field theory, JHEP 04 (2016) 091, [1601.02883].
- [31] J. Estes, K. Jensen, A. O'Bannon, E. Tsatis and T. Wrase, On Holographic Defect Entropy, JHEP 05 (2014) 084, [1403.6475].
- [32] D. Gaiotto, Boundary F-maximization, 1403.8052.
- [33] T. Nishioka and I. Yaakov, Supersymmetric Rényi Entropy, JHEP 10 (2013) 155, [1306.2958].
- [34] A. Kapustin, B. Willett and I. Yaakov, Exact results for supersymmetric abelian vortex loops in 2+1 dimensions, JHEP 06 (2013) 099, [1211.2861].
- [35] T. Okuda, Mirror symmetry and the flavor vortex operator in two dimensions, JHEP 10 (2015) 174, [1508.07179].

- [36] K. Hosomichi, S. Lee and T. Okuda, Supersymmetric vortex defects in two dimensions, JHEP 01 (2018) 033, [1705.10623].
- [37] L. Anderson and M. M. Roberts, Supersymmetric space-time symmetry breaking sources, 1912.08961.
- [38] N. Hama and K. Hosomichi, Seiberg-Witten Theories on Ellipsoids, JHEP 09 (2012) 033, [1206.6359].
- [39] V. Pestun, Localization of gauge theory on a four-sphere and supersymmetric Wilson loops, Commun. Math. Phys. 313 (2012) 71–129, [0712.2824].
- [40] J. M. Maldacena, The Large N limit of superconformal field theories and supergravity, Int. J. Theor. Phys. 38 (1999) 1113–1133, [hep-th/9711200].
- [41] E. D'Hoker, J. Estes and M. Gutperle, Exact half-BPS Type IIB interface solutions. I. Local solution and supersymmetric Janus, JHEP 06 (2007) 021, [0705.0022].
- [42] B. Assel, J. Estes and M. Yamazaki, Large N Free Energy of $3d \mathcal{N} = 4$ SCFTs and AdS_4/CFT_3 , JHEP **09** (2012) 074, [1206.2920].
- [43] D. Bak, A. Gustavsson and S.-J. Rey, Conformal Janus on Euclidean Sphere, JHEP 12 (2016) 025, [1605.00857].
- [44] M. Gutperle and A. Trivella, Note on entanglement entropy and regularization in holographic interface theories, Phys. Rev. D95 (2017) 066009, [1611.07595].
- [45] S. Ryu and T. Takayanagi, Holographic derivation of entanglement entropy from AdS/CFT, Phys. Rev. Lett. 96 (2006) 181602, [hep-th/0603001].
- [46] T. Takayanagi, Holographic Dual of BCFT, Phys. Rev. Lett. 107 (2011) 101602, [1105.5165].
- [47] S. Ryu and T. Takayanagi, Aspects of Holographic Entanglement Entropy, JHEP 08 (2006) 045, [hep-th/0605073].

- [48] K. Jensen and A. O'Bannon, Holography, Entanglement Entropy, and Conformal Field Theories with Boundaries or Defects, Phys. Rev. D88 (2013) 106006, [1309.4523].
- [49] E. Lauria and A. Van Proeyen, $\mathcal{N} = 2$ Supergravity in D = 4, 5, 6Dimensions, vol. 966. Springer, 2020, 10.1007/978-3-030-33757-5.
- [50] P. Breitenlohner and M. F. Sohnius, An Almost Simple Off-shell Version of SU(2) Poincare Supergravity, Nucl. Phys. B178 (1981) 151–176.
- [51] M. de Roo, J. W. van Holten, B. de Wit and A. Van Proeyen, *Chiral Superfields in* $\mathcal{N} = 2$ *Supergravity, Nucl. Phys.* **B173** (1980) 175–188.
- [52] D. Butter, B. de Wit, S. M. Kuzenko and I. Lodato, New higher-derivative invariants in $\mathcal{N} = 2$ supergravity and the Gauss-Bonnet term, JHEP **12** (2013) 062, [1307.6546].
- [53] B. de Wit, J. W. van Holten and A. Van Proeyen, Structure of $\mathcal{N} = 2$ Supergravity, Nucl. Phys. **B184** (1981) 77.
- [54] A. W. Fisher, $\mathcal{N} = 2$ Supersymmetric Yang-Mills Theory and Its Supercurrent, Nucl. Phys. **B229** (1983) 142–156.
- [55] J. Gomis, P.-S. Hsin, Z. Komargodski, A. Schwimmer, N. Seiberg and S. Theisen, Anomalies, Conformal Manifolds, and Spheres, JHEP 03 (2016) 022, [1509.08511].
- [56] K. Hori, S. Katz, A. Klemm, R. Pandharipande, R. Thomas, C. Vafa et al., *Mirror symmetry*, vol. 1 of *Clay mathematics monographs*. AMS, Providence, USA, 2003.
- [57] S. Sugishita and S. Terashima, Exact Results in Supersymmetric Field Theories on Manifolds with Boundaries, JHEP 11 (2013) 021, [1308.1973].
- [58] D. Honda and T. Okuda, Exact results for boundaries and domain walls in 2d supersymmetric theories, JHEP 09 (2015) 140, [1308.2217].
- [59] K. Hori and M. Romo, Exact Results In Two-Dimensional (2,2) Supersymmetric Gauge Theories With Boundary, 1308.2438.